PP

-

National Criminal Justice Reference Service



This microfiche was produced from documents received for inclusion in the NCJRS data base. Since NCJRS cannot exercise control over the physical condition of the documents submitted, the individual frame quality will vary. The resolution chart on this frame may be used to evaluate the document quality.



MICROCOPY RESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS-1963-A

Microfilming procedures used to create this fiche comply with the standards set forth in 41CFR 101-11.504.

Points of view or opinions stated in this document are those of the author(s) and do not represent the official position or policies of the U. S. Department of Justice.

National Institute of Justice J United States Department of Justice Washington, D. C. 20531 Date Filmed

3/27/81



PAROLE AND PROBATION

NYSIIS

RECEIVED MANACEMENT SET

· • · ·

20

STUDY



FINAL REPORT

Bureau of Systems Development New York State Identification and Intelligence System Albany, New York

3

Contributors

NEW YORK STATE CRIMINAL JUSTICE AGENCIES

Representatives of:

Corrections Courts District Attorneys Parole Police Probation

NYSIIS Survey Team Members

Bureau of Systems Development

John F. Burbige Lester A. Kipp Gary E. Schreivogl

Bureau of Systems Planning and Research

James F. Moon Thomas H. Mullen Alexander Shayne

Report Preparation

NYSIIS John F. Burbige

Computer Usage Corporation

Nicholas Carozza

PREFACE

3

.

This report is one of a series of publications designed to disseminate the results of the various special projects conducted in association with Project SEARCH. Most of these special projects were undertaken because there were potential technical or operational implications of significant interest to all participating states. However, the projects were conducted by individual states and primarily addressed intra-state problems.

In approving the publication of these special project reports, the SEARCH Project Group is attempting primarily to accomplish wide dissemination of all of the results coming from SEARCH-associated efforts. It should not be assumed, in these special project reports, that recommendations and conclusions presented by the authors are endorsed by either the SEARCH participants or by LEAA. Rather, the SEARCH Project Group believes that this report is of sufficient quality and comprehensiveness that it deserves consideration in any similar project.

ii

Approved for publication

0.J. Hawkins Chairman SEARCH Project Group

APPENDIX

GLOSSARY

)

)

4

)

Section

TABLE OF CONTENTS

Title	Page
Introduction	1
Information Needs	4 *
Background and Major Functions of New York State Parole and Probation	18
The NYSIIS Response Type-Distribution and Priorities	51
Data Element Require- ments of the Parole/ Probation Reporting System	71
Parole/Probation Reporting System	79
Criminal Justice Statistical Reporting System	92
	100
	101
	103
	105
	106
	109

LIST OF TABLES

<u>Table</u>	<u>Title</u>	Page		
2-1	Police Decision Points and Required Systems Assistance			
2-2	District Attorney Decision Points and Required Systems Assistance	ALL TABLES		Chart
2-3	Judicial Decision Points and Required Systems Assistance	EXCEPT 5-1 Are	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	6-1
2_4	Probation Decision Points and Required System Assistance	GROUPED IN LAST		6-2
2-5A	Local Correction Decision Points and Required Systems Assistance	SECTION OF REPORT		6-3
2 - 5B	State Correction Decision Points and Required Systems Assistance			
2-6	Parole Decision Points and Required System Assistance			
2-7	Volume Loads and Response Time Needs			
4-1	Type Response By Agency Decision Point			
5-1	NYSIIS Present and Planned Data Base Elements; Source and Use	74-78		

).

.

3

1

iv

2

LIST OF CHARTS

v

<u>Title</u>	Page
NYSIIS Status File Operations By Major Justice System Decision Points	82
NYSIIS Inquiry Process	90
NYSIIS Update Process	91

1

SECTION I INTRODUCTION

GENERAL INFORMATION

2

1

8

P

This is the final report of the initial Parole and Probation Study conducted by the New York State Identification and Intelligence System (NYSIIS). Briefly, the goals of the Parole and Probation Study were to survey and document the information needs of parole and probation agencies and also to define the methods wherein parole/probation information could be made rapidly available to the criminal justice community. With the accomplishment of these goals, NYSIIS will have the capability to proceed with the design and implementation of an advanced parole and probation reporting system, permitting participating agencies to contribute to and share in the information stored in the system.

SURVEY OBJECTIVES

In order to accomplish the stated goals of the Parole and Probation study, the project team, organized for this effort, conducted the survey in accordance with the following objectives:

 Determine what information is presently available within parole and probation agencies that can be made readily accessible to the criminal justice community through NYSIIS.

	2.	Deter
		parol
		satis
	3.	Deter
		from
		in sa
· · · · · ·		proba
	4.	Deter
		from
		in sa
		proba
	5.	Deter
		estab
		acces
		in th
		proje
	6.	Deter
		each
		batio
		inter
INFORMA	rion	SOURC
The	e ini	format
primari]	ly-fı	rom tw

4

)

. 3. . .

5

1 1

٤

)

1)

3 3

1

3 🐌

)

The information contained in this report was derived primarily-from two sources. The first source consisted of field trips made by members of the project team to representative parole/probation agencies, courts, police agencies, dis-

mine the specific information needs of e and probation agencies that can be fied by NYSIIS.

-2-

rmine what information is presently available, other agency sources, that would be of use atisfying the information needs of parole/ ation agencies.

omine what information is presently available, other agency sources, that would be of use atisfying the information needs of parole/ ation agencies.

rmine those data elements required for the blishment of a statistical data base to be ssed by contributing agencies for assistance he performance of program evaluations and ections.

rmine the most efficient method of integrating agencies indicated needs for parole and proon information, into an automated basic data schange system.

CES

trict attorney offices, and correction institutions within New York State. The data collected on each field trip was then documented in a report written by a member of the project team in attendance at the agency interview. All avail-, able prior reports pertinent to the current study and individual agency publications were also reviewed for additional information.

-3-

FORMAT OF THE REPORT

R

.

2

.

L

e

£

E

The following sections of this report present a detailed analysis of the results of the investigation performed by the NYSIIS Parole and Probation Study team. Section II is devoted to an analysis of the basic information needs of criminal justice agencies with emphasis given to the needs of parole and probation agencies; section III provides a brief background on parole and probation in New York State and analyzes these operations through an examination of their major functional areas; section IV describes the type and frequency of NYSIIS assistance to the system agencies; section V discusses the data element requirements of the NYSIIS reporting system; section VI describes the revisions to the NYSIIS process system necessitated by the survey results and recent legislative action and section VII provides for an initial description of the planned NYSIIS statistical system.

INTRODUCTION

)

1

£

The key area of information sharing between criminal justice agencies concerns the continuing relationship of an offender to the entire criminal justice process from entrance to exit. When was he arrested, and with what offenses was he charged? What disposition did the court make in his case? Was he placed on probation or was he committed to an institution? Was he later paroled? Has he had a continuing record of criminal activity, with multiple arrests, convictions and sentences? In general, the question at issue is--what is known about the criminal background of a given individual? All agencies are concerned with these questions. The contents of this section identify the relevant information needs of criminal justice agencies at their specific decision making points as surveyed by the Parole/Probation study team.

The types of agencies interviewed by the project team fall into six distinct groups: probation, parole, police and sheriff, courts, district attorneys, and penal institutions. A defined procedure was used by the project team in interviewing each agency that represented one of these six groups. This procedure generally took the following form: 1. Explain the proposed role and objectives of the Project SEARCH Parole and Probation study in terms of a centralized parole/probation reporting system and the criminal justice community.

SECTION II INFORMATION NEEDS

-4-

2. Define the major agency decision points in processing criminal offenders and obtain the corresponding information needs at each of these decision points. In the same context, obtain agency response time and volumes, attitudes and factors relevant to the enrichment of the NYSIIS SCH (Summary Case History) data base particularly in the area of parole and probation.

-5-

POLICE NEEDS

C

Police functions essentially consist of two major decision points: investigation and arrest/booking. As shown on table 2-1, police needs for parole/probation information at these decision points are limited. However, the police would like to be notified as soon as possible if an arrestee is on parole or . probation so that they may cause him to be detained pending appropriate parole or probation action.

Additionally, the arrest agencies expressed a desire to be notified of all parolees and probationers under supervision in their particular area, and the conditions of their release. The police feel that in some instances, this information will provide assistance during investigative work and additionally could identify an otherwise seemingly minor infraction of law as a violation of a condition of parole or probation.

DISTRICT ATTORNEY NEEDS

E,

Ci

3

 \cap

0

The District Attorney offices visited by the survey project team indicated that their primary concern in the investigations functions involved sensitive issues, such as political corruption and organized crime. All other investigations were performed by local criminal justice agencies. However, the agencies visited stated that background information was necessary in prosecutions, particularly in the bail and ROR phases of a case.

-6-

The major District Attorney decision points are: arraignment, pre-hearing, grand jury, trial and sentencing. Table 2-2 lists these decision points with the corresponding operations, systems assistance and type response. It is evident from the table that the District Attorney must have access to current or updated parole/probation information concerning the criminal offender. Specifically, a central reporting system can assist the District Attorney in three ways: 1. The District Attorney requires parole/probation and "Wanted" status information before bail/ROR precodings. 2. The District Attorney requires parole or probation information prior to plea.

3. The District Attorney requires accurate and complete information concerning the suspects prior criminal and social history.

COURT NEEDS

1

S

T

е

е

£

C

C

3

L

C

R

To determine the courts' parole/probation information needs at their key decision making points, the project team interviewed administrative staff members of the New York City Criminal Courts. Table 2-3 lists the decision points with the corresponding operations, systems assistance and type response.

-7-

In order to assist a criminal court in bail and ROR proceedings, a central reporting system must possess the ability to provide background (in and out of state) criminal history information in a timely fashion. In addition, background information is needed prior to sentencing, especially in those cases where there is a possibility of a sentence of probation. Such information would greatly assist the presiding judge in decision considerations. In regard to the pre-arraignment phase (Bail - ROR) it was indicated by court officials that an abbreviated version of the current NYSIIS criminal history would be quite sufficient at this stage of the judicial process primarily due to the short time period available for arraignment. It was suggested by the project team that a response of this type should include the following information so as to best satisfy the criminal courts operations:

1. A record of the current arrest.

2. A listing of all previous criminal history both in and out of state. This could be in tabular format and in chronological order. 3. A record of any as a warning de by the court's
4. A synopsis of the clude general is military record cotics history.
PROBATION NEEDS
In surveying the in project team determined activities were pre-array fer, supervision, violat

C

 \cap

In surveying the information of probation agencies, the project team determined that the key decision making points or activities were pre-arraignment, pre-plea, pre-sentence, transfer, supervision, violation, discharge and follow-up. Table 2-4 lists these decision points with their corresponding operations, required system assistance (information needs) and type of response.

The overall reaction to the proposed inclusion of probation data in a central reporting system was enthusiastic. All of the probation agencies interviewed saw it as a significant advance towards a full criminal justice information system. A set of data elements (see Attachment A) was utilized by the survey team as a guideline during each probation agency interview. Each probation agency commented in a positive manner upon these items and generally accepted them as the prime requisites of a probation reporting system.

3. A record of any mental hygiene institution admissions as a warning device. This would permit further study by the court's psychiatric staff.

4. A synopsis of the individual's social record, to include general information such as previous addresses, military record, marital status and alcohol and narcotics history.

1.1

-8-

Throughout the survey each probation agency was asked to comment on what additional information it deemed pertinent to an extended reporting system. The following are those major additions to the system suggested to the survey project team.

- 1. Establishment of a "County/State of Record" probation file which would include the name and supervising jurisdictions of all persons currently on probation in the state.
- 2. Increased reporting, on the part of probation agencies, to the central files so as to facilitate the inclusion of pertinent probation data.

r

r

¢

t

ç

e

ę

ç

(C

ćC

۲.

- 3. Initiation of a reciprocal system between individual criminal justice agencies and the central state agency for reporting deaths of individuals of interest to the system. This procedure could, for example, be formulated as follows:
 - a. If the death of an individual is reported to NYSIIS and his records indicate he has had contact with another justice agency (e.g. probation department), NYSIIS would notify the probation department of the death. This would allow the department and NYSIIS a means for file purging.
 - b. If a probationer under the department's supervision dies, this information would be reported to NYSIIS to allow purging of its files.
- 4. The inclusion of Mental Hygiene, Narcotic Addiction, Alcoholism and co-defendant information would be extremely valuable. However, prior to the approval of

release of this type information a valid method of verifying such information must be resolved. 5. A method wherein investigating probation agencies could ascertain the existance of prior probation

-10-

CORRECTION INSTITUTION NEEDS

 \bigcirc

()

 \bigcirc

٢

O

()

 \mathbf{O}

0

 \odot

0

A. New York City Department of Correction The survey team interviewed officials of the New York City Department of Correction and the New York City Reception Center for men at Rikers Island to determine Correction information needs at their major decision points in the processing of inmates. The City Correction Department was interested in the possibility of systematically including parole and probation information in the NYSIIS reporting system.

The following is a summary of this data:

-9-

reports, their location, type and availability.

There are five major decision points in the correction process as determined by the survey project team. These are: detention, post-sentence (commitment), transfer, release and follow-up. Table 2-5A lists these decision points with the corresponding operations, systems assistance and type response.

1. DETENTION: Currently, the New York City Department of Correction receives criminal history information from the New York City Police Department at the time of detention. This history generally includes city-wide fingerprintable arrest and available dispositions and usually excludes non-fingerprintable and non-city arrest and disposition information. To completely meet their information

needs at detention, Correction officials request that NYSIIS provide a chronological synopsis of an individual's prior criminal record. Minimally, this record should include the following data elements:

-11-

Name NYSIIS Number Charge(s) Crime Date(s) Disposition(s)

.,F

Arresting Agency(s) Sentence(s) Mental Hygiene Admission(s) Narcotics Status

Also, if attainable, the reason for the mental institution admission would be valuable to assist in reducing the possibilities of injuries to and suicides among inmates. As to priorities, criminal history background information on individuals arrested for lesser offenses (non-fingerprintable) is considered highly important in that the individual is more likely to be released before the background information is available. Ideally, this information should accompany the individual to the detention facility.

2. POST-SENTENCE: Correction requires the inmate's entire NYSIIS criminal history record at this decision point to assist in the classification process. Here, as in detention, the Correction officials classify the need for background information concerning individuals convicted of lesser offenses as one of their more prominent needs. This is due primarily to the daily receipt rate at the classification center, (170/day),

 \cap

0

O

æ.

٢

٢

0

 \bigcirc

٢

Ĉ,

 \mathbf{O}

with the result that the Department may only allow three days for classification processing prior to transfer of these inmates to one of the permanent incarceration facilities spread throughout the city and state. In addition since most of these inmates have been convicted of non-fingerprintable crimes. NYSIIS has not been informed of the arrest and consequently has not previously been able to release a criminal history to the institution. This lack of information, however, will be greatly reduced beginning September 1, 1971, due to the expanded category of fingerprintable crimes in New York State (CPL 160.10) and to the increased number of occasions where a pre-sentence report will be required (CPL 390.10; 390.20). TRANSFER: The transfer and subsequent location of inmates within the institution is viewed as a serious problem to Correction officials. The concern has become so great that the Department is presently considering automation of its inmate record keeping system. Compounding the problem of transfer within the institutions of the City Correctional system is a recent agreement

between City and State Correction officials to permit the City to transfer portions of its longterm inmate population to State Correctional institutions to ease the overcrowding conditions of City institutions. RELEASE: Correction needs a more comprehensive interchange of data in this area. For example, on occasion

-12-

the courts will call for someone who is no longer in the custody of the Department. Significantly, Correction does not notify or provide exit data (discharge, escape, death) to NYSIIS. However, the general feeling was that Correction could provide the following data, if requested:

-13-

Name NYSIIS Number Charge Arrest Date Place of Detention Sentence Release Date Type of Release

5. FOLLOW-UP: The Correction officials interviewed expressed interest in the potential of NYSIIS for providing statistical information which would be of use to them in program evaluation. A variety of possible report types were discussed, (see Table 2-5A - Follow-Up Data), and both NYSIIS and the Correction officials agreed to investigate further the possibilities of utilizing central agency data in facilitating this evaluation.

B. New York State Department of Correction

The needs and requirements of the State Department of Correction are, with the exception of detention, significantly similar to the needs of the City Department. Response times are not as critical as the City Department due to the longer sentence length and more stable inmate population. Table 2-5B lists these decision points with the corresponding operations, systems assistance and type response.

PAROLE NEEDS

(

61

0

Ð

1

 \mathbf{O}

0

()

(

Due to the centralized structure of the New York State Parole function, the survey team concentrated its efforts on a series of interviews with the Central Offices of the Division of Parole. The result of the interviews was an agreement, as in probation, to investigate the parole function via its major decision point responsibilities.

There are seven major decision-making points in the parole process, as determined by the participants. These are hearing, release to parole, transfer, general supervision, violation, discharge and follow-up. Table 2-6 lists these decision points with the corresponding operations, required system assistance and NYSIIS response. The following is a detailed summary of this data for the Division of Parole:

1. HEARING: A report is prepared at this point by the Institution Parole Officer at the reception center showing the social, marital, military, educational, etc. background of the potential parolee. In general, the preparation of this report amounts to updating the pre-sentence Probation Report. If additional information is needed it is acquired via the local area parole office. As far as can be determined, a central parole/probation reporting system, will not usually be called upon to aid Parole at this decision point.

-14-

2. RELEASE TO PAROLE: Parole is given advance notice when an inmate will be released and to which Area Parole Office he will be assigned. The information required at this point is contained in the Parole case folder received from the institution. If any information is missing a request for it is made through the Department of Correction. System assistance is not needed at this decision point.

UC

:C

1

₫€

1

1

Э

1

10

C

C

-15-

- TRANSFER: Transfers are handled by the transferring 3. and receiving agencies. Copies of transfer reports are forwarded to central parole headquarters by both agencies. System assistance at this point is dependent upon the type of transfer (in or out of state).
- GENERAL SUPERVISION: There are three categories for 4. supervision: intensive, active and reduced. Under intensive supervision, the parolee must report from 2-4 times a month, for active it is once a month, and under reduced it is 4 times a year. Supervision information is required at this decision point to show social rehabilitation of the parolee. This type of information is presently collected by the Division of Parole's Research Unit.
- 5. VIOLATION: There are three categories of violations that parole is concerned with: criminal, technical and absconding. At this decision point, parole wants

()

C .

 $\mathbb{C}_{\mathbb{N}}$

٢

٢

(2)

٢

Ð

 \bigcirc

-16-

to be notified on all arrests if the person is currently on parole. This is one of the areas where a central parole/probation reporting system can be of assistance. It is standard practice for Parole to notify NYSIIS of all parole violators so that the NYSIIS files can be updated. When a warrant is issued, the Area Parole Office notifies the Central Parole Office. The Central Parole Office then notifies NYSIIS, who posts this information to its WANTED FILE and N.C.I.C. Once entered into the NYSIIS files, NYSIIS will be prepared to notify Parole as soon as an activity is recorded on the parolee.

6. DISCHARGE: At this decision point a parolee is discharged when his maximum time is up. According to parole authorities, there is no system assistance needed at this decision point.

7. FOLLOW-UP: At present, follow-up information is available to Parole through statistical studies undertaken and published by the Research Unit of the Division of Parole. The basic system for parole does not, due to the sophistication of the Parole Departments Research function, suggest any system assistance at this stage, other than normal criminal history reporting.

In summary, a central parole/probation reporting system can aid parole by reporting all persons arrested who are currently under parole supervision. Parole, can assist in the formulation of a true and complete criminal justice reporting system, by supplying NYSIIS with a greater range of information (perhaps via the Department of Correctional Services) in a more timely fashion.

VOLUME LOADS AND RESPONSE TIME REQUIREMENTS

' C

Ĉ

Table 2-7 summarizes the case load volumes and required response time of the various types of agencies interviewed by the parole/probation project team. This data is given for each agency's key decision points.

a C. of th el C PA pr 0 Ne be th \bigcirc or 0'n Sei \bigcirc Jai of wii \bigcirc Det 0 whi iza cha ٩ pos ren 0

This section of the report will concern itself with both a review of the functions of parole and probation as segments of the criminal justice system and an examination of each of these functions via an analysis of the major and supportive elements of each function.

PAROLE

 \mathcal{O}

Parole may be defined as the conditional release of a prisoner serving an indeterminate or unexpired sentence. In New York State, the application of this process has, by law, been vested in a Board of Parole in the Division of Parole of the State's Executive Department (Correction Law 210). This organizational structure, however, has recently been modified. On May 8, 1970, the Governor signed into law New York State Senate bill number 9086-A which, when it goes into effect on January 1, 1971, will remove both the State Board and Division of Parole from the Executive Department and consolidate it with the New York State Department of Correction into a single Department of Correctional Services.

This legislation is the latest in a series of revisions which relate to modifications of the operational and organizational characteristics of those New York State agencies charged with the rehabilitation and supervision of pre and post adjudicated offenders. It is not the intention of this report to present a detailed study of all these prior revisions.

-17-

-18-

SECTION III

BACKGROUND AND MAJOR FUNCTIONS OF NEW YORK STATE PAROLE AND PROBATION

However, this report will discuss one other recent innovation which, due to its wide ranging impact, does significantly affect the overall responsibilities of the parole function in New York State.

C,

€

· C

Effective September 1, 1967, by legislative enactment, the New York State Division of Parole assumed the state-wide responsibility of providing parole services to all individuals sentenced to a term of imprisonment in any institution in the State of New York. Pricr to this date there had been two independent parole authorities in the State, the New York City Parole Commission responsible for the parole of all individuals incarcerated in institutions under the control of the New York City Department of Correction and the New York State Board of Parole whose responsibility covered all inmates incarcerated in State penal institutions.

In brief, this legislation provided for the absorption, by the State Division of Parole, of the personnel of the New York City Parole Commission and all parolees supervised by this Commission. In conjunction with this assumption of state-wide parole responsibility, the State Division of Parole was also vested with the responsibility for implementing and operating a newly conceived program for providing full parole services to those inmates designated as lesser criminal offenders (i.e. serving sentences of one year or less). This program, the Conditional Release Program, also became effective on 1 September 1967 and with it New York initiated a system of insitution release (parole) for those individuals receiving a definite

sentence of imprisonment of between ninety days and one year. Under the provisions of this program, inmates are eligible for release after they have served sixty days (including jail time) of that term. Effective April 1, 1971, the requirements for consideration will be modified to pertain to sentences of between sixty days and one year, and eligibility after thirty days service (no jail time).

Release, in either case, interrupts the sentence and the remainder is held in abeyance. Should the individual violate the conditions of his release he may be returned to the institution to serve the remainder of his original sentence. Any individual accepting conditional release is subject to supervision as follows:

for a one year period.

se

Un

re

of

CO

s1

se

re

th

sť

Ari

รน่

ye

1n

me

opi

1.

C

52

 \odot

 \mathbf{O}

The results of the agency consolidation process, while not as yet completely known, will provide a high yield of positive benefits in the area of parole practice and operation. One of the most immediate gains is the emergence of a unified state-wide parole operation, supervised by a single Board of Parole. Addition-

1. If at release the unserved portion of the term is not in excess of 120 days, after allowing for jail time and good behavior time, the inmate must accept supervision

2. If the unserved portion is in excess of 120 days the supervision period will be two years.

-20-

ally, the consolidation permits state-wide standards and procedures to be implemented and applied to all inmates who qualify regardless of the type penal institution.

Perhaps the single most notable benefit, from an information reporting system viewpoint, is the formation of a central records section which contains the parole files of all New York State parolees and parolees supervised by New York State for other states. This centralized filing system substantially assists both the "record keeping" process of the Division and the Interagency Reporting system recently initiated between the Division of Parole and NYSIIS.

MAJOR PAROLE FUNCTIONS

C

The parole process may be considered as consisting of the three major functional areas of investigation, supervision and research. Each of these functions while seemingly performed in an independent mode is nevertheless closely entwined with each of the other functions and additionally with the entire criminal justice process.

A. Investigation

The parole investigation function is a continuing process which commences upon the receipt of a sentenced individual at a classification center, continues throughout his incarceration and release period and in most cases terminates upon release from legal custody. (In some cases the investigative process may continue into the research function). The primary purpose of the investigative function is to provide information concerning the background of the sentenced

offender which will assist correction and parole authorities in the planning and implementation of institutional and supervisory programs which offer the best prospects for rehabilitation. This investigative function may be subdivided into three types of supporting functions, namely the classification investigation, the parole investigation and general investigations.

1. Classification Investigation As soon as possible following the receipt of a sentenced offender at a correctional classification center, the institutional parole office begins the classification process. This entails a thorough review of the presentence report performed by a probation agency; an interview with the prisoner and if necessary requests for further background investigative assistance through a parole office. The objectives of the classification investigation are basically two-fold.

-to initiate the professional casework relationship between inmate and officer; a relationship which will eventually continue between parolee and officer and; -to gather, analyze and evaluate all available background information on the inmate to help determine the institution and program assignment which seems to offer the inmate the best opportunity for rehabilitation. In order to accomplish a successful classification investigation, the institutional parole officer must have the facilities to acquire timely and pertinent data on the inmate. While this acquisition process has required

-21-

-22-

3

 \bigcirc

 \mathcal{O}

ès.

 \bigcirc

t-

C

 C_{1}

 \bigcirc

O

0

considerable effort in the past, it has been of lesser concern recently due to the increase in the number of pre-sentence probation reports transmitted to the classification center. This increased transmission and use of pre-sentence probation reports has, in addition to providing significant assistance in the data acquisition process, also assisted parole in its establishment of positive casework relationships by allowing more time for inmate-officer interviews and discussions.

-23-

CENTRAL AGENCY SYSTEM ASSISTANCE

€

¢,

C

C

C

C

C

C

Since one of the most vital ingredients in a successful classification investigation is a comprehensive pre-sentence probation report, NYSIIS contends that by supplying detailed background information to probation authorities, it can most expeditiously assist Parole in achieving its classification objectives. This process of supplying information to probation authorities will be covered at length in the probation portion of the report. However, it must be emphasized that if a parole officer requires assistance from NYSIIS during the classification process, the assistance should be readily available.

2. Parole Investigation

This investigation is basically similar to the classification investigation discussed above and is prepared by the institutional parole officer for use by the Parole Board in examining inmates for release consideration. Although similar to the classification investigation, it differs to the following degree -It contains information relating to the inmate's conduct while in custody, the degree to which the inmate did or did not apply himself to the rehabilitation process, and his record of educational and/or vocational progress, etc.

-It chronicles the attitude of the inmate as it was revealed through the supplementary meetings between himself and the parole officer during the incarceration period. These periodic supplementary sessions permit the parole officer to assess the inmate's performance to date, and give the inmate the opportunity to comment on his progress and his attitude towards his potential for success upon release. This report also indicates the type employment, if any, the inmate tends to pursue upon release.

CENTRAL AGENCY ASSISTANCE

٢

C

)IC)

Q

;iđ)

st

O

C

C

 \bigcirc

rees \bigcirc

lle

(1

CEN

at

to

or

Hoï

pa:

3.

There does not seem to be any need for agency assistance at this point, since NYSIIS will have already replied either to the probation department in its pre-sentence investigation or to the parole officer in his classification investigation. However, as in the classification process, NYSIIS will be prepared, if called upon, to render history assistance at this point.

- 3. General Investigations cooperative investigations.

In 1969, parole officers in New York State were called upon to perform in excess of 14,000 investigations. Of this total close to 6,000 investigations were performed in areas, which for the purposes of this report, will be classified as general investigations. These are classified here as: residence-employment, good conduct, executive clemency and

a. Residence-Employment Investigations

This type of investigation is performed in order to evaluate the proposed residence and employment plans

-24-

of an inmate. The residence portion of this investigation requires that the investigating officer gather information relating to both the type of residence and the individuals residing at the location. In those cases where the prospective residence is other than the inmates familial residence, the investigation may need to determine the presence of prior criminal associates or co-defendants. (This need is also recognized but to a lesser extent, if the prospective residence is in the inmates familial residence.) In complying with the requirements of the employment portion of this investigation, the officer will request assistance in securing similar information on the potential employer and/or other employees.

-25-

CENTRAL AGENCY ASSISTANCE

. C. .

. C

15

,C

C

¹C

C

A central agency could offer assistance to the investigating officer by supplying him, upon request, with information which will aid him in his attempts to divert the parolee from relationships (prior associates or individuals with prior criminal experience) or places which may hinder his opportunities for a successful parole period.

b. Good Conduct Investigation

The purpose of this investigation is to provide, within the limits prescribed by law, for the restoration of some or all of the civil rights or privileges lost by certain convicted offenders. This investigation is performed by the Division of Parole for not only those individuals who were under parole supervision but additioninstitution releases. of the following:

bn

bn

()

 \mathcal{O}

 \odot

O

O

 \mathbf{O}

O

0

 \mathbf{O}

O

The type of investigation performed will depend on whether or not the individual under consideration has ever been under parole supervision. The investigation performed on individuals with no prior parole supervision experience follows closely the requirement of the previously discussed Classification report. In those cases where the applicant has prior parole supervision experience and a classification investigation has been performed, this current investigation will supplement it by gathering information on the period since his discharge from parole supervision.

ally, upon the request of those who received conditional release (suspended) sentences, fines, or maximum expiration institution releases.

-26-

The criteria for eligibility are varied and in general consist of the following:

- Applicant must have five years of acceptable conduct from the date of conditional release sentence, date of payment of fine or date of final institution release.

Applicant, regardless of time lapse from date of parole, may not be considered if currently in a parole supervision status.
Applicant, if parole violator, may not be considered until five years have lapsed from the date of reparole.

- Applicant, if cause was out-of-state conviction, must be an inhabitant of New York State for at least five consecutive years prior to application.

The investigation is performed by an area office investigator and his report is submitted to the Chairman of the Board of Parole for consideration. The application is reviewed and acted upon by three members of the Board and unanimous approval is required for granting the relief certificate.

CENTRAL AGENCY ASSISTANCE

C

C

C

С

ſ

0

C

In rendering assistance to the parole investigator performing the Good Conduct investigation, NYSIIS could provide two distinct types of responses. In those instances, where there has been no prior supervision experience, NYSIIS would respond with a comprehensive summary case history on the individual, as in the classification process. In those cases where there is a prior classification report available, parole would indicate this fact and NYSIIS would reply with a chronological recapitulation of the applicant's criminal activities, if any, from the date of his release from supervision.

-27-

c. Executive Clemency Investigation

The Constitution of the State of New York empowers the Governor to grant reprieves; commute sentences and issue pardons following conviction for all but a small number of specific offenses. Applications for Executive Clemency are presented to the Office of the Governor, who at his discretion, forwards the request to the Board of Parole for investigation. The procedure followed in investigating Executive Clemency requests is very similar to the procedure utilized in the Good Conduct Investigation and the investigative scope of the report follows the Classification investigation format.

CENTRAL AGENCY ASSISTANCE

 \mathbb{C}^{1}

 \bigcirc

 \mathbf{D}

Û

 \mathcal{O}

0

0

The type assistance rendered at this point would, due to the similarities between this investigation and the Good Conduct investigation be similar to that indicated in the Good Conduct portion of this section.

d. Cooperative Investigations

The basis for this type investigation is found in the application of the Interstate Parole Compact, to which all fifty states are signatories. A cooperative investigation is required in two instances.

- When an offender from an out-of-state correctional institution, eligible for parole, indicates that he would prefer supervision in New York State.

- When an individual with a New York State criminal record is confined in another state, that state may request an investigation so as to better understand and evaluate the individual.

In either of the above instances, the investigation will generally conform to the needs and requirements of the classification investigation.

CENTRAL AGENCY ASSISTANCE

C

C

C

C

 \mathbf{C}

In investigating the requests for out-of-state parolees to reside in New York State, the parole officer will initiate his investigation by requesting a New York State criminal record if any and by investigating the potential residence and employment possibilities of the individual. In satisfying the request of the other states for prior New York State history, NYSIIS would respond, via the State Division of Parole, with a comprehensive summary case history or norecord response.

Implications of Investigative Assistance

The benefits which would accrue to the parole function and the investigative process due to the assistance described thus far would seem to be significant in a number of ways. In the first instance the assistance is recognized as satisfying much of the background informational needs of the investigation. In addition, the information is standardized and would present a uniformity of response not otherwise easily attainable if solicited from divergent sources. And finally by providing for the identification of other potential information sources, the response will considerably reduce the amount of time the officer must expend locating these sources thereby permitting more of his time to be allocated to the critical parolee-officer counseling and supervisory function. B. Supervision

 $\langle c \rangle$

 \mathbf{C}

O

C

0

C

ted

CI

 $\mathbf{\hat{n}}$

0

The parole supervision function is primarily concerned with offering guidance, assistance and counsel to the parolee. This supervisory period commences at the moment the parolee arrives for his initial interview with his supervising parole officer and officially continues until he satisfactorily completes his parole period or is returned to an institution as a parole violator. Unofficially, parole officers often find themselves rendering assistance to prior parolees far beyond the culmination of the parole period.

In the performance of the daily supervisory process, the parole officer must possess the data necessary to monitor and guide the parolee's progress. Much of the investigative work has already been performed, however, it should be emphasized that the supervisory officer may, for any number of reasons, require immediate additional information on a wide range of matters. While most of these additional informational needs are not easily predictable or categorized, there do exist areas such as the identification of new potentially dangerous associations and the identification of newly arrested individuals as parolees where the supervising officer requires information as a matter of routine and a central information agency could offer significant assistance. The manner in which the information relating to associates would be transmitted to the operating agency is similar to the transmission method which will be discussed in the Probation portion of this section.

-29-

-30-

In the area of identification of possible parole violators. the central agency would examine all incoming arrest records received from any area in the state, and by scanning an active parole file, quickly identify the arrestee as a parolee and notify both the central office of parole and the arresting authorities as to the arrest and status of the individual. This is current procedure between NYSIIS and the State Division of Parole. Without this procedure, the Division of Parole would have to resort to its previous method of determining whether or not a parolee had been arrested: assign a parole officer to a daily review of arrest listings at police or sheriff's agencies. This review consumed a considerable number of parole-officer man-hours, and affected the total supervisory process by keeping otherwise qualified officers involved in nonsupervisory functions. While this process has as yet not been discarded, the foundation for its removal has been laid in the newly implemented reporting system. As this system expands in use and sophistication, it can be assumed that the manual daily review requirements will correspondingly decrease and eventually become unnecessary.

-31-

C

1

C:

Implications of Supervisory Assistance

By providing timely and routine assistance to the parole supervisor, a central agency can also benefit parole in two indirect but essential ways, namely, increased utilization of qualified parole officers in the supervisory process and increased value of the supervisory period to the parolee by permitting a close parolee-officer involvement.

-32-

C. Research

 \mathfrak{O}

Ó

C

C

٢

٢

O

3

.

۲

The utilization of sound research techniques in the analysis and evaluation of programs has long been recognized by Parole as a prime requisite for the efficient application of the parole process. In 1957 the Division of Parole initiated a Bureau of Research and charged it with the responsibility for conducting research projects and operational analyses of various parole activities and operations. This Bureau has undertaken a number of research projects which have contributed to a better understanding of the motivations and personality characteristics of offenders. The studies performed by this unit reflect more than the usual basic "head-count" type of research. Instead, many of these research and statistical studies are analytically oriented and reflect the ways in which the relationships of basic social, economic, educational advantages, or lack thereof, contribute toward the increase of criminality in society today. It is through research studies such as these that the criminal justice system will be able to evaluate more realistically the effects of its policies and practices on the offender as he passes through the system.

For example, once documented and formulated, basic relationship research could be expanded to demonstrate the results of various sentences on certain types of offenders (i.e. where probation is ultimately more valuable to society in regard to certain offender types than a period of confinement); or the effects of varying degrees of counseling aids (intensive-moderatenome) on similar and diverse groups of parolees. It is anticipated that these types of studies would provide the groundwork for the eventual reduction of the unknown factors contributing to initial and persistent criminal behavior.

-33-

CENTRAL AGENCY ASSISTANCE

I,

C.

C

1

A central agency can contribute to this type of research in a variety of ways. First, since it is a central repository for the basic system data, it is in a unique position to supply this data for advanced research studies. It can also assist a research function by removing the necessity for duplicate data collection, hence permitting it to spend more time on the research-development phases of a project. Another way it may offer assistance is by producing the research studies once they have been fully developed thereby permitting additional research projects to be undertaken. Lastly, through the knowledge and operational experience gained, it would seem feasible for a central agency to produce composite agency statistical reports which would reflect the total offender treatment process. PROBATION

()

 ϵ

C.

Ô

C

0

()

()

12

 \mathbf{O}

()

Probation is defined as "the action of suspending the sentence of a convicted offender and giving him freedom during good behavior under the supervision of a probation officer". /1 This definition, however, renders little understanding of the complete probation process. Indeed the definition delineates but one of the three basic elements of the probation function which are investigation, supervision and research.

The responsibility of probation has grown from its initial concept as a function which supervised the conduct of a convicted offender whose sentence had been suspended to one which generally becomes involved in all investigative matters pending before the court it serves. The original probation emphasis was upon pre-sentence investigation and field supervision of crime and delinquency but as courts became increasingly involved in social services (i.e. family court, juvenile proceeding, adoptions) and the corresponding information needs, these investigations became the concern of the court's probation agency. Coupled with this increase in 'non-criminal' investigative areas was an increase in the need for probation to perform additional 'criminal' investigative operations in the pre-adjudicatory phase of proceedings (bail - ROR, etc.) These factors (combined

1 Websters Seventh New Collegiate Dictionary

-34-

with a relatively rapid increase in the crime and arrest rate have contributed greatly to the formidable workload found in most New York State probation agencies.

te

6

2

:0

U

<

C

In order to fulfill these diverse needs and requirements, there exists sixty-nine distinct and relatively autonomous probation departments in the State of New York. These departments are by and large structured at the county level, and depending on the size of the department may service either or both the family and criminal court functions. In certain counties, one probation department provides all the services for all the courts of the county (civil as well as criminal). Other counties distinguish probation responsibility between criminal court services and family court services. In the state's largest city, there exists four distinct probation departments -- one servicing all family courts and courts of inferior jurisdiction and three others servicing the needs of the city's five Supreme courts. This multiplicity of probation responsibility and obligation is also found, via specialized type service, within each probation department division. The combination of all these factors makes difficult any single universal definition of the probation function. These operations run the broad length and breadth of judicial responsibility and encompass operations ranging from the performance of bail and presentence investigations to family court investigation and from supervising convicted felons to supervising wayward minors.

Probation in New York State was formally recognized in 1901, when the first probation law relating to persons over the age of 16, was passed. In 1905 a Probation Commission was appointed to study probation in the State and as a result of the findings of this commission, the State Probation . Commission was established in 1907. In 1927 as a result of the findings of the Wickersham Commission, New York State established a Division of Probation in the State Department of Correction. This organization of the Division of Probation remained in effect until May 8, 1970 when the Governor signed into law Senate bill number 9097 which removed the Division from the Department of Correction and established it as a Division of the State Executive Department effective January 1, 1971. Since its inception in 1907 the Division of Probation has grown from an agency which supervised 1,672 offenders using 35 probation officers to an agency which in 1968 supervised 63,693 offenders under the direction of 1774 probation officers.

The New York State Division of Probation is charged with the responsibility of supplying general supervision over the administration of probation throughout the state. The Division is empowered to collect statistical and other information and is charged with the duty to secure the effective application of the probation system and the enforcement of the probation laws.

-35-

-36-

0

Ô

()

11

0

 \mathbf{O}

0

(O)

These mandates, as set forth in the Correction Laws of the State of New York, clearly outline the major functions and responsibilities of the Division of Probation. However, the enforcement of these provisions in relation to local probation operations poses problems to the Division. Many local probation administrators are responsible to four distinct sources of supervision and are vague as to what phase or phases of their operations are controlled by each supervisory phase. The four sources are the county government, the New York State Judicial Conference, the State Division of Probation and the particular justice of the court being served. It is this diversity and multiplicity of operation which contributes to the most recent evaluation by the New York State Division of Probation of the operations of Probation in New York State. This evaluation is based on the findings of the Governor's Special Committee on Criminal Offenders and observes that "as a result of the diversity and multiplicity of these semiautonomous units, probation services could not be evaluated as good, bad or indifferent. Members of the Committee also could not evaluate whether probation supervision in case work, as presently administered, is helping offenders". /1

-37-

MAJOR PROBATION FUNCTION

E

C

C

Ċ

1

()

C

C

C

As mentioned previously, probation consists of the three major operational areas of investigation, supervision and research. This section of the report will, as in the parole section, examine each of these major areas and the ways in which

/1 Annual Report for 1968 - NYS Division of Probation

a central reporting system may assist them.

INVESTIGATIVE

10

()

()

()

()

()

()

 \bigcirc

0

If any function of probation may be considered as the cornerstone of that process, or perhaps the criminal justice system, it would seem to be this investigative function. The importance of performing a proper investigation cannot be over-emphasized due to the heavy impact the investigation results will have on the subsequent phases of the offender's system involvement. There are several types of probation investigation, but all are performed in the context of satisfying two basic criteria:

-acquiring all available and pertinent information concerning the individual in order to provide the ingredients necessary for proper evaluation of prior socio-criminal experience and to facilitate the prudent application of various alternatives (e.g. bail, ROR, sentence of probation, etc.)

-providing the basis for planning institutional or supervisory programs which seem to offer the maximum benefits to the offender and the community upon the offenders exit of the system.

-38-

Types of Investigations

C

C

Ċ

 \mathbf{O}

0

In discussing the various types of investigations performed, this report will limit the discussion to those concerned with the criminal process.

-39-

BAIL - RECOGNIZANCE (ROR) INVESTIGATIONS

The investigation is performed in order to provide the court with information and thus, relative to the offender's prior history, facilitate the prudent application of the judicial options available at this stage of the proceedings. These investigations must be performed expeditiously without sacrificing any information or accuracy requirements. The present procedure requires that a probation officer interview the offender to acquire data relating to prior criminal activity and present criminal status, employment and educational experience, residence information, etc. The officer then attempts to validate, mainly through telephone calls, as much of the information as possible. Based on the results of the interview and the validation procedure the officer recommends to the presiding judicial officer the procedure deemed appropriate (i.e. bail or recognizance release or detention). The performance of this investigation, however, is restricted in varying degrees, due primarily to the severe time limitation often imposed on the investigator. Secondly, much of the required information is scattered throughout the files and records of various criminal and social agencies. Many times a probation officer will have no awareness of prior criminal activity

bation report.

C)

 (\circ)

 \cap

 \bigcirc

ty

t()

 $\left| \left< \right> \right>$

 $\langle \cdot \rangle$

()

0

CENTRAL AGENCY ASSISTANCE

In order to facilitate this investigation, NYSIIS would be prepared to transmit to the investigating officer an abbreviated summary case history indicating all known prior in-and out-of-state judicial information. In addition, this response would include, if available, previous employment experience and residence information. By providing this type of data in an expeditious manner, NYSIIS will significantly reduce the time presently required to locate and retrieve pertinent data, thereby permitting the officer to expend more time, if necessary, in the data verification process. An additional benefit of this central agency involvement is that the timely acquisition of accurate data would release more of the officer's time for use at the supervisory level.

Pre-Plea Investigations

In performing this type of investigation the probation officer requires essentially the same kinds of information as in the bail-ROR investigation. The investigation is basically performed to satisfy one or both of the needs below:

-to provide information necessary in the post-indictment period in regard to the advisability of bail or recognizance release;

outside his given area and unless this information is offered by the offender, it will not be included in the pro-

-40-

-to provide information to assist the court in its determination of whether or not to accept a plea to a lesser offense.

-41-

CENTRAL AGENCY ASSISTANCE

C

1C

C

C

C

 \bigcirc

In offering assistance here, a central agency should be prepared to render the same type of response as that described in the proceeding bail-ROR discussion. One significant advantage at this point in comparison to bail investigation is a larger time frame for responses.

The advantages of central agency assistance here are substantially the same as the advantages indicated under the bail description; namely, a wider scope of data, reduction of required investigation time and increased availability of supervisory time.

Pre-Sentence Investigations

This investigation is performed, in order to provide the court with detailed information concerning the defendants prior record and the current offense. It is a report upon which the court relies heavily in passing sentence. In basic format the report is similar, insofar as acquisition and correlation of data are concerned, to the two previously discussed types of investigations. However, the preparation and presentation of this report dictates that the investigating officer expend a considerable amount of additional time in developing the background of the offender via a detailed ex-

planation of the prior and current social climate of the individual and an analysis of the ways, if any, these conditions have precipitated the present situation.

-42-

In gathering this type of information the probation officer has a number of resources upon which he will call to complete this investigation. He may have the prior probation reports on the individual, and he should have all the support documents which permitted the compilation of these reports. In addition, the officer will expend considerable time and effort visiting the family and area of residence of the defendant, visiting relatives, clergymen and social and business associates, finally he will interview individuals who have known the defendant in the educational and employment areas. All of these visits and others deemed relevant, will assist the officer in portraying a true and complete background report of the offender.

CENTRAL AGENCY ASSISTANCE

0

1

 (\cdot)

()

O

 \mathbf{O}

 \mathbf{O}

 \bigcirc

0

0

A central agency reporting system would render significant assistance to the investigator by supplying him with a complete background history of the individual. The history released at this point in the system would differ from those released at the two previous system points in that it will supply information such as the location, type and availability of previously performed probation investigations. This criminal history entry will greatly facilitate the investigation, by providing the officer with an awareness of the existence of prior probation investigative data and thereby, where practical, reduce the time expended in what often amounts to repetitive

investigations. This reduction in investigative time will also render significant benefits to the supervision function.

-43-

TRANSFER INVESTIGATIONS

The administration of the probation function provides, when just cause exists, the means whereby probationers may apply for inter- or intra-state relocation. This procedure usually entails the completion of a transfer investigation by the receiving agency before it will accept responsibility for the probationer.

Intra-State Transfer

C

 \mathbf{C}

1 C

-<u>C</u>

O

Upon application, the probationer is required to give evidence that the change in location will enhance his potential for probation success. The officer will then evaluate the merits of a transfer, and if he concludes that the transfer would be beneficial to both the community and the probationer, he will forward an official request to the prospective supervising agency for permission to effect this change. At this point, the prospective office will perform a transfer investigation, using as its focal point the original pre-sentence report and the supervision reports compiled on the probationer up to this point. In most cases this information is sufficient for the investigation. Occasionally, however, the prospective agency may request that an up-to-date background history be provided, prior to its acceptance of the individual. When the change in supervision is effected, the receiving agency would notify the

central agency, so that in the event of future fingerprintable activities by the individual, the central agency will be able to quickly notify the proper supervisory agency.

-44-

Inter-State Transfers

 \bigcirc

 \bigcirc

 \bigcirc

 \mathbf{O}

0

 (\mathbf{i})

C

0

O

0

t

Investigations to effect this type transfer concern the New York State probation function only when the individual requests a transfer into New York State. However, they concern NYSIIS whenever individuals request permission to transfer supervision either into or out of the state as the central agency may be called upon for information on transferees in either direction.

When an individual requests permission to transfer his supervision from New York State to another state the procedure followed (examination for just cause, forwarding pre-sentence reports, etc.) is essentially the same as described in the preceding Intra-State procedure. In addition, the central agency should be prepared, as above, to furnish the prospective outof-state supervisory agency with an up-to-date background report if the new agency deems it necessary. At present, these requests for further information would be channeled through the original New York State probation agency to the central agency and the responses would be returned in a similar fashion. If the transfer is approved, it would be the responsibility of the original New York State probation agency to notify the central agency of the new location and agency identification number of the individual.

In those cases wherein individuals request permission to transfer supervision into New York State, the prospective New York State agency will request that a complete background investigation be performed. This investigation will be assisted by the central agency by providing any prior history it has on the individual. As in the previous instances, the new agency of supervision should be duly registered in the central system to facilitate, should the need arise, the central agency notifying the proper supervisory agency in cases of alleged violations of probation.

Other Uses of the Probation Investigation

C

έÇ.

13

C

The examination of the investigative probation function has thus far indicated its use by the courts (in bail-sentence proceedings) and by subsequent probation responsibilities (supervision, other probation reports). However, the utilization of the results of these investigations extends far beyond the aforementioned and affects the eventual operations of the correction function and the parole function. In the correction area the results and evaluations contained in the presentence report will provide the basis necessary for proper classification of the offender at the reception center and should provide assistance in the appropriate placement of the offender in the type of correctional program which offers him the best opportunity for advance. The probation reports also play an important role in the entire parole function. Initial parole contact with the probation report is in the classification process where as previously discussed, it is used as a basis for the assignment of the offender to a type of training. Based on the conclusions and recommendations contained in the report parole prepares a supervision program for the individual.

SUPERVISION

1

13

()

 $(\)$

 C_1

h

1

C

O

Cu

Gr

The purpose of supervision in the probation system is to provide, for those deemed qualified by the court, the opportunities within a community environment which will offer the offender the maximum rehabilitative potential. In the performance of his duties, the supervising probation officer is dependant on a number of various factors. The success or failure of the supervisory function depends a great deal on the recommendations set forth in the prior investigative process, for it is here, as previously discussed, that the probationer's background, motivations and aspirations were first uncovered and evaluated, and based on these findings a supervisory program was suggested. The function also depends on the attitude of the probationer: his capacity and his willingness to conform to regulatory procedures and the degree to which he applies himself to the self-improvement process. Another factor in the supervision period is the attitude of the community, especially the attitude of those with whom the probationer is most

-45-

-46-

likely to associate. And finally, the success or failure of the supervisory function depends, to a large extent, upon the amount of time a probation officer has available to provide guidance and counsel for the probationer.

Embodied among supervisory duties is the necessity for a constant awareness, on the part of the probation officer, of the current status of the probationer. This awareness is maintained not only through interviews with the probationer, but also by visits to his family, his employer and associates. Keeping track of associates, particularly new ones, is seen in conjunction with the need for timely notification of new arrest data as the two most immediate supervisory information needs.

CENTRAL AGENCY ASSISTANCE

Ċ

C

C

The central agency can assist in satisfying these needs by forwarding upon request any background information in the file on newly acquired associates, and could supply the rapid notification required in the identification of new arrestees as probationers. Such a response could immediately be transmitted to the probation office listed as the supervising office and to the arrest agency currently holding the individual. While this notification is deemed necessary for all arrests, it becomes a matter of particular concern when the arrest is based on a relatively minor infraction of law and the entire system process (arrest-arraignment-trial-sentence of probation or fine) may be completed before any agency becomes aware of the individual's current status. Since the investigative process continues throughout the protation period, the officer must allocate portions of his available supervisory time to the investigative process. Here, as before, the central agency can assist by rendering informational services thereby permitting for more time to be expended in a purely supervisory environment.

RESEARCH

e

Ħ

1

 C_{λ}

 (\cdot)

 C^{+}

 \mathbf{O}

 \mathbf{O}

Ô

O

f

S

r

The utilization of sound research techniques has long been recognized as necessary if effective probation procedures are to be realized. However, research techniques in the probation function have not been developed to any large degree on a state-wide basis. The underlying causes for this lack of development seem to be the result of: - the autonomy of the various probation agencies; - the diverse levels and types of supervisory control; - the number of agencies composing the probation function; - the demand for operational services (investigationsupervision) of probation which precludes the assignment of research related studies.

While these factors hinder, in varying degrees, the refinement of data evaluation and research techniques on the state level, the survey team did find some research efforts relating to individual agency operations in some of the agencies surveyed.

-47-

-48-

For example, on one of its visits the team had occasion to meet with and discuss the operations of an agency's Division of Research. This Division is actively engaged in on-going research projects on a county basis, and has initiated, among others, a study relating to the evaluation of probation and its effect on recidivism. /1 It is anticipated that the study when completed will provide the agency with the information it requires to evaluate its present and planned probation programs and the relationships these programs have to the rate of recidivism. In the carrying out of this study, the research team anticipates the assistance of NYSIIS in the gathering of research data concerning a sample number of individuals who have been under the jurisdiction of this agency. In recent years this agency has also published the results of studies relating to the effects of its Parole and Recognizance program and an analysis of all individuals arrested on drug related crimes. /2

-49-

The New York City Office of Probation has undertaken a number of studies relating to evaluation of its operation. Many of these studies have been in conjunction with the VERA Institute of Justice and one of the most recent analyses published was a comprehensive examination of and recommendations for the increased use of the Release on own Recognizance

"AN EVALUATIVE RESEARCH STUDY OF PROBATION IN NASSAU COUNTY AND ITS EFFECT ON RECIDIVISM" NASSAU COUNTY PROBATION DEPARTMENT-DIVISION OF RESEARCH.

NASSAU COUNTY PROBATION DEPARTMENT ANNUAL REPORT 1967.

5

program in the criminal courts. /1

 \bigcirc

4

15

 (γ)

÷.

VT.

O

 \bigcirc

There are undoubtedly other research projects underway throughout the state, and the impression perceived by the survey team indicated the growing awareness on the part of the surveyed agencies of the benefits which would accrue to those agencies which vigorously pursued this research and evaluative method. In addition, the assurance that NYSIIS would be prepared to offer systems assistance in the pursuit of their research goals is recognized by these agencies as a valuable assist in the formulation of such programs.

-50-

NYSIIS is likewise prepared to offer assistance to the New York State Division of Probation in the undertaking of research programs designed to measure and evaluate the results of probation on the state level. While the nature of assistance has not as yet been fully defined, NYSIIS anticipates pursuing the potential for such programs with the Division of Probation once its organizational change is effected on January 1, 1971. The type of assistance which NYSIIS would provide would be designated at both the state and county levels and would closely parallel the type of assistance previously discussed in the parole portion of this section.

January 1969

The PROBLEM OF OVERCROWDING IN THE DETENTION INSTITUTIONS OF NEW YORK CITY-AN ANALYSIS OF CAUSES AND RECOMMENDATIONS FOR ALLEVIATION - ANDREW SCHAFFER VERA INSTITUTE OF JUSTICE

SECTION IV THE NYSIIS RESPONSE: TYPE - DISTRIBUTION AND PRIORITIES

-51-

The preceding section of this report has detailed the operational needs of parole and probation in New York State and outlined the methods whereby NYSIIS may render functional assistance. This section of the report will concern itself with a recapitulation of the present reporting system and a discussion of the types and distribution of responses that will satisfy these basic needs. The conclusions in this report in regard to type and distribution of agency assistance are based on two underlying factors:

> -the stated needs and requirements for parole and probation information exhibited by all surveyed agencies.

-the increased responsibility placed on the operational agencies of the system and NYSIIS, by the recently enacted New York State Criminal Procedure Law (CPL).

PRESENT NYSIIS INFORMATION REPORTING SYSTEM

C

€.

5

 $^{\circ}C$.

C

 \mathbf{C}

- C

€

At present, NYSIIS produces two versions of its criminal history response. These responses are classified as manual and computer and within each there are "no record" responses and full criminal history responses. The manual response is a tabular report and is prepared as a reply to arrest or inquiry transactions concerning individuals whose records, because they did not meet the original computerization criteria /1 have not yet been computerized. This manual response is used only once since the receipt of the current transaction, if it is a criminal fingerprint input, automati-

cally qualifies the individual's record for computer conversion. However, since the time required for conversion will vary depending on the individual's prior history, NYSIIS satisfies its reporting requirement by releasing either a full manual history or a manual "no record" response prior to computer conversion. Any transactions pertaining to this individual subsequent to the one which precipitated conversion will be responded to by a computerized criminal history. In the event the current transaction is other than a criminal fingerprint input, however, (e.g. license application, employment investigation, etc.), the individual's record will not be converted.

The computerized criminal history response (attachment B-1) is released to requesting agencies if the individual is currently on the computer or if the individual is a "first-timer" (i.e. no prior record). This response will reflect all prior information on the individual including the current event. In the case of "first-timer" status the computerized response will indicate both the no prior record condition and the current event. This procedure in computerized "no record" responses is followed only in those instances where the current input is a criminal fingerprint.

responses directly, via facsimile or mail, to the input agencies. REVISED NYSIIS INFORMATION REPORTING SYSTEM As a result of the conferences between the SEARCH project team and the criminal justice agencies surveyed and as a result of the increase in reporting responsibility designated to NYSIIS by the new CPL, NYSIIS undertook an examination and evaluation of its present reporting system in an attempt to determine its adaptability towards fulfilling the future information dissemination needs of NYSIIS. This portion of the

Current distribution procedures call for NYSIIS to forward its

-52-

re

th

ne

pr

wi

th

I.

()

 $| \cap$

0

 \bigcirc

 \mathbf{O}

Ľ

^{/1} The basic criteria utilized in the initial computer selection process required that in order to qualify for conversion an individual must have been born in 1912 or later and must have been arrested at least once between 1947 and 1967 (1967 was year of conversion).

report will concern itself with the results, as known thus far, of this internal examination and evaluation.

-53-

In order to both effectively satisfy the diverse informational needs of the various agencies in the criminal justice system and to provide this data in a manner which will permit its utmost use, NYSIIS will concentrate on the production and dissemination of histories in the following ways:

I. TYPE AND PURPOSE OF THE RESPONSE

C

 \mathbf{C}

(C ...

be

1

·B--

C

ie.

e

on

t 1s

C

 \bigcirc

and

Şe

IIS

tem

C

he

rd)

A. COMPREHENSIVE SUMMARY CASE HISTORY REPORT

This report, (see attachment B-1) as furnished by NYSIIS, will contain an account of all the information on file with NYSIIS concerning the individual. It will be an extension on the present criminal history response containing in addition to the current information, those data elements not presently collected by NYSIIS but deemed essentual for a full report by the agencies surveyed. This additional data supplied to the system by the agencies, is seen as consisting of bail, ROR information, parole/probation supervision data, supervision transfer information, location, type and availability of prior pre-sentence reports. It is anticipated that the system will also permit the future collection of verified information relating to educational and employment experience, military data, birth date and place data and a number of other relevant social characteristics. This type of response will be transmitted to fulfill the two areas of the justice system wherein full background data is deemed necessary: the probation presentence investigation area and the police crime investigation phase. Each of these areas requires extensive backfunction.

Ō

0

 C_1

l

 \mathbf{O}

1

 \mathbf{C}

O

 \odot

This NYSIIS response (see attachment B-2) is a concise, tabular recount of the individual's prior history. It contains all the pertinent data displayed on the full report but without the detail found thereon. Briefly, the report will be composed of six sections as follows:

3.

ground information on an individual, or a list of suspects if it is to complete successfully its investigative

B. ABBREVIATED SUMMARY CASE HISTORY REPORT

-54-

1. SUBJECT IDENTIFICATION SECTION

Consisting of name-State Identification number; FBI number (if available); also known as; race; skin tone; sex; date of birth; height; and social security number.

2. CURRENT STATUS SECTION

Contains a statement of the current criminal justice status of the individual such as on parole/probation; under sentence of conditional discharge; wanted and either the supervisory agency and charge for which supervised, or the wanting agency.

ARREST AND JUDICIAL DATA SECTION

Containing arrest date, arrest agency, local agency ID number, arrest charge, latest judicial action date, disposition charge, and decision. The arrest and judicial segments of this section will be associated via arrest date.

POST-ADJUDICATORY INFORMATION SECTION 4. Reflects sentence information and contains final judicial date, sentence, term or length of supervision, and institution or supervision area.

-55-

5. MENTAL HYGIENE INFORMATION SECTION

Indicates admission date, patient identification number, discharge date and name and location of institution for individuals admitted as a result of a criminal adjudication.

OTHER INFORMATION SECTION 6.

Ċ

10

1

10

5

C

C

C

Ci

C

C

Contains specific information in tabular format under sub-headings such as, Employment (date, firm, location, position); Education; Military (date(s), branch, type discharge, service no.); Addresses (location, frequency); Marital Status; Dependants; Narcotic/Alcohol user; Associates (names-dates and ID numbers, if known).

This response will be the type that will be most often produced and distributed by NYSIIS. It will be used for supplying information to: courts, and probation agencies in the preliminary stages of judicial action (Bail-RORpre-plea); police agencies, who according to law must receive a copy of the criminal record; and district attorneys who require background information in carrying out their duties in these preliminary stages.

HIGH PRIORITY REPORT C.

Whenever the processing of an arrest record indicates that the subject is presently registered with the system as a parolee, probationer, or fugitive, NYSIIS in addition to

producing the normal report, will also produce a notice for transmission to the supervising or the wanting agency indicating to that agency that the subject has come under the official jurisdiction of another agency. This notification will contain elements (i.e. name, agency ID, present agency and location, etc.) sufficient to permit identification of the subject (see attachment B-3).

D. NO FURTHER ACTIVITY OR SUBSEQUENT ACTIVITY RESPONSE Occasionally, agencies will have a need for gathering information pertinent to an individual's activity from the date of the last report, such as when a considerable length of time passed between the probation pre-sentence report, (at which point NYSIIS has already replied), and the actual date for sentence. The court may wish to confirm that there has been no further criminal activity in this time period. In processing this request, NYSIIS would scan its files; if no further activity were encountered, a response indicating this fact would be released. If activity were encountered, NYSIIS would reply with a subsequent report beginning where the previous report terminated. Another potential user of this type response would be a parole classification officer, under conditions similar to those described above.

II.

 \cap

 \mathbf{O}

0

 C_i

CI

 \mathbf{O}

 \bigcirc

, [

DISTRIBUTION AND DERIVATION OF THE CRIMINAL HISTORY

The accompanying table (table 4-1) displays, via agency and agency decision points, the type response which NYSIIS will transmit to the indicated agencies to facilitate fulfillment of their mandates. The provisions of the new Criminal Procedure Law, as they affect the recording and dissemination of criminal history and background information on individuals are included and shown where

-56-

applicable.

0

3-

1

 $\langle \cdot \rangle$

C

30

C

 $\langle 0 \rangle$

CE

icy

es.

C

A. POLICE AGENCY

The activities of police agencies generally fall under two types of operation: investigation and arrest/booking. NYSIIS will be prepared to offer assistance at each of these points.

-57-

- 1. <u>INVESTIGATION</u> (table 4-1; coordinates A-1) NYSIIS will be prepared to assist the police agencies in their investigation of crime. Upon receipt of the name or names (and SID number when available) of suspects in a criminal case, NYSIIS will search its files and if the suspect(s) have prior criminal activity, a comprehensive summary case history will be released. In the event the suspect is unknown, a "no record" response will be transmitted. In addition, NYSIIS will be able to assist this investigative phase by reporting to the police agency those individuals whose present status preclude their being considered as suspects, thereby reducing the workload of the police investigator. For example, should the individual be currently serving a sentence of imprisonment or should he be listed as dead, this information would be reported.
- 2. <u>ARREST/BOOKING</u> (table 4-1 coordinates B-2) Immediately upon arrest, the arresting agency is required to notify NYSIIS of this event if the arrest charge is a fingerprintable offense (section 160.10 of the recently enacted Criminal Procedure Law greatly expanded the category of fingerprintable arrests by adding many new offenses to this category). Upon receipt NYSIIS will

nea	ss and	a
a.	Fing	er
	The	ne
	for	a :
	to b	a1]
	crim	ina
	his j	pri
	its (cor
	requ	Lre
	a cor)y
	finge	erp
	furni	sh
	agend	y
	In or	de
	polic	e,
	the t	ур
	and w	
		1.
		2.

 \hat{C}

0

CI

11

0

Thus the following will hold true in the NYSIIS processing and dissemination of information at this point:

review all arrest fingerprint input records for completeness and accuracy and will respond as follows:

printable Offenses

-58-

w CPL stipulates that an individual arrested fingerprintable offense may not be admitted all or recognizance release until the local hal court has received from NYSIIS a copy of for history, if any, or no record response, for insideration (CPL 530.20). A further stipulation res that the District Attorney's office receive of the defendant's prior history, in all printable instances, and that this copy be hed to the District Attorney via the arrest (CPL 160.40).

er to satisfy the diverse obligations of prosecutor, courts (probation) and NYSIIS, be response and response method will vary

1 be predicated upon two basic considerations: The results of the NYSIIS field survey of parole/probation (court) and other

agencies. The requirements of the CPL affecting

the operations and obligations of each agency.

URBAN AREAS

-for every arrest, NYSIIS will prepare two copies of an abbreviated summary case history and will transmit both copies to the arrest agency. The arrest agency will retain one copy for its files and will forward the second copy to the appropriate district attorney.

- for every arrest NYSIIS will prepare two copies of an abbreviated summary case history and will transmit these copies to the indicated court and the probation agency.

RURAL AREAS

r

on

C

1

:

 \bigcirc

51

 \mathbb{C}

C

- The timely distribution of history information to "upstate" New York areas poses a problem of considerable magnitude. Due to the wide diversity of local criminal courts which may be called upon to participate in this type of bail or recognizance proceeding (there are over 2500 justice courts involved) NYSIIS, without further study of potential transmission possibilities, cannot at this time concretely describe the best dissemination method. However, when this communication study is completed, NYSIIS will be in a position to determine the best method of satisfying the needs of both the surveyed agencies and the mandates of the new law. Until such time, NYSIIS will transmit abbreviated summary case history information to the arresting agency and will request that they provide for the prompt dissemination of the information to the agencies concerned.

b. Other Response If, during the processing of any arrest fingerprint. the subject's record indicates that he is Wanted, under post-adjudicatory supervision (parole-probation-conditional discharge), or in a preliminary supervisory status (bail-recognizance), NYSIIS will immediately notify both the agency exercising jurisdiction and the arresting agency informing them of his current location and system status through the utilization of the high priority response. B. DISTRICT ATTORNEY

-60-

The needs of the prosecutor relating to the awareness of background information on an individual will be satisfied via, responses by NYSIIS to inquiries and updates of police and probation agencies.

For example, although NYSIIS defined four basic system decision points concerning district attorneys, it was suggested that in only the Preliminary Hearing (Initial Arraignment) and Arraignment (plea) phases could NYSIIS be of significant assistance. The remaining points of trial and sentence do not appear to require central agency responses (a possible exception is recognized when during the course of a trial a prosecutor may request additional information concerning a previously unknown witness or event).

1. PRELIMINARY HEARING (table 4-1 coodinates C-1) NYSIIS will, as outlined in II-A-2-a above (page 58) forward an abbreviated summary case history to the arresting police agency so that the police agency may transmit it to the prosecutor's office.

-59-

 \mathbf{O}

(1)

 \cap

O

O

 \mathcal{O}

0

O

0

2. ARRAIGNMENT (table 4-1 coordinates E-1) Subsequent to the return of a felony indictment by a Grand Jury the court myst again consider the bail/ROR possibilities and permit the district attorney to be heard. Hence, at this point it may become necessary for NYSIIS to forward additional copies of the criminal record to both the court and prosecutor. In those instances where the defendant has remained in custody throughout the preceding period, the criminal history initially forward to the court and prosecutor will suffice. In the event the defendant was previously released on recognizance or admitted to bail prior to indictment, the prosecutor and court may require an additional central agency response. For this NYSIIS will reply to the arraignment court with two copies of either the no further activity response or the subsequent report response. One copy of the report will be for the use of the district attorney and the other for the use of the court and probation agency.

-61-

COURT C.

C

°C

C

C

se.

C

ia

C

sion 0

in

ign-

e. C

ay

C

C

The four general decision making areas are Preliminary Hearing (Initial Arraignment); Arraignment (post-indictment); Trial, and sentence. Of the four only the trial phase is seen as not requiring any direct central agency assistance.

1. PRELIMINARY HEARING (table 4-1 coordinates G-1) The information needs here will be satisfied by the NYSIIS response to the input arrest document as described in II-A-2-a above (page 58).

below (page 63).

2.

()

 $(\mathbf{5})$

()

 $(\mathbf{0})$

 \mathbf{O}

0

ŝ

- 3.
- D. PROBATION AGENCIES

The major decision points investigated by the team involve bail/ROR investigation, pre-plea and pre-sentence investigations; probation supervision and termination of probation.

1.

-62-

ARRAIGNMENT (table 4-1 coordinates H-1) The information needs will be satisfied by the NYSIIS response to the probation inquiry outlined in II-D-2

<u>SENTENCE</u> (table 4-1 coordinates J-1) Information needed by judicial agencies will be satisfied by the NYSIIS response to the probation pre-sentence inquiry outlined in II-D-3 below (page 63).

The NYSIIS survey of probation agencies was oriented towards an investigation of the needs and requirements of these agencies. In addition the team ascertained the kinds of data possessed by these agencies and their availability for dissemination. The new CPL has been particularly cognizant of probation function needs and has taken steps to define more fully the role of probation with the remainder of the justice community. NYSIIS, based on the survey results and the new CPL requirements is also cognizant of its obligation to provide active support to the probation function.

BAIL/ROR INVESTIGATIONS (table 4-1 coordinates K-1) The new CPL stipulates the increased areas of investigative responsibility of probation agencies in the bail/ROR process. NYSIIS will assist these agencies by responding as indicated previously in the Police Agency portion of this section (II-A-2-a page 58).

2. PRE-PLEA INVESTIGATIONS (table 4-1 coordinates L-1) NYSIIS will assist the probation agencies at this stage of the judicial proceedings by responding with a subsequent report. This report will also assist both the court and the prosecutor. Information received here is seen as decidedly affecting the judicial process in the "plea" stage.

-63-

Ţ

1

1

ť

ye

C

- PRE-SENTENCE INVESTIGATIONS (table 4-1 coordinates M-1) 3. Section 390.10 of the CPL stipulates that the court may not pass sentence on any fingerprintable offense conviction until it has received a pre-sentence report from NYSIIS. This report will contain a comprehensive history of the defendant and will be utilized by the probation agency in the preparation of the pre-sentence report for the court.
- SUPERVISION (table 4-1 coordinates N-1) 4. If during the course of probation supervision, the probationer has contact with a police agency, this information will be transmitted to the supervising probation agency as outlined in II-A-2-b above (page 60). In addition, NYSIIS will be prepared to provide a probation agency with histories at any time the agency deems it necessary.
- 5. <u>RELEASE FROM SUPERVISION</u> (table 4-1 coordinates 0-1) Prior to release from supervision, a supervisory office may require an up-to-date background history on the probationer. In this event, NYSIIS will respond with an abbreviated criminal history to the inquiring probation agency.

Once an individual is released from supervision, for whatever reason, NYSIIS will be so informed and will update both the basic records and the status file records of the individual.

-64-

E. PAROLE The major decision point functions of the parole process, as they relate to a central agency information, may be listed as pre-parole investigation, release to parole, supervision requirements and release from parole investigations.

1. <u>CLASSIFICATION INVESTIGATION</u> (table 4-1 coordinates P-1) This investigation is performed by the institution parole officer and consists of preparing a complete background resume of the inmate for submission to the parole board. Since the CPL requires that the pre-sentence report prepared by the court probation staff be forwarded (along with all source reports) with the prisoner to the person in charge of the correctional facility (CPL 390.60), it is not likely that NYSIIS will be accessed for such information.

RELEASE TO PAROLE (table 4-1 coordinates Q-1) 2. Central Files of the Division of Parole will notify NYSIIS of all individuals released on parole so that the latters file may be updated.

3.

0

11

O

()

O

 \bigcirc

 \bigcirc

 \mathbf{O}

SUPERVISION (table 4-1 coordinates R-1)

NYSIIS will respond to the Central Parole Files with a notification anytime an active parolee, registered as such with NYSIIS, is arrested and that arrest is reported to NYSIIS. In addition, NYSIIS will prominently display on
its reply to the arresting agency, prosecutor, probation department and court involved; the criminal justice status of the individual (see II-A-2-b page 60). As in the probation supervision phase, NYSIIS will be prepared to forward additional reports to parole upon request.

 \bigcirc

 \mathbf{O}

(1)

 $|\langle \cdot \rangle|$

()

 (\cdot)

Ci

 \cap

()

-65-

RELEASE FROM PAROLE (table 4-1 coordinates S-1) Upon the successful completion of parole period or upon re-incarceration, at board direction, the Division of Parole will inform NYSIIS of the action so that the NYSIIS files may be updated.

F. CORRECTIONS

Ć

Ċ

ť.

S

 \mathbf{C}

C

h

C

The survey team examined Corrections functions within the framework of their relationship with the parole and probation aspects of the system and with emphasis on what was deemed Corrections four significant function decision points. These four points are detention; receipt; visitors; and release. Each of these either contributes to or benefits from an active parole-probation/central agency relationship.

1. <u>DETENTION</u> (table 4-1 coordinates T-1)

Upon receipt of an individual for detention, those charged with this responsibility often find themselves faced with a grevious problem. In effect this problem concerns the frequent lack of background data on the individual. The absence of this data often prohibits the facility from employing the security necessary to insure the safety of both the detainee and the other residents of the facility (e.g. if the facility were aware of a detainees' prior mental or narcotic problems it would assign the individual

to a psychiatric or medical ward, rather than a general cellblock). In providing this information NYSIIS will routinely produce a copy of its criminal history for transmission to the County Sheriff's office in "upstate" (non-New York City) New York State locals. The dissemination of history information for detention purposes in New York City, however, represents an area of considerable difficulty to both NYSIIS and the New York City Department of Correction. Estimates of detainees by this department put this detention figure at between 60,000 and 75,000 individuals per year. Background information on these individuals is required in most cases within a matter of hours following arrest. In order to resolve this difficulty, NYSIIS is exploring two alternative methods of dissemination: the first of these would request that one of the agencies which initially received the arrest-originated criminal history from NYSIIS, forward its copy, or a copy of the report to the detention facility with the individual. The second possibility would require that the New York City Department of Correction request a history on all individuals remanded to their custody directly from NYSIIS. While the latter alternative is recognized as the most direct approach it is none-the-less the more difficult due primarily to the formidable volume involved. 2. ADMISSION (table 4-1 coordinates U-1) As a result of the CPL requirements that the pre-sentence

III.

-66-

report accompanying the offender to prison there is no need for formal reporting procedures at this point. However, NYSIIS will be prepared to forward a comprehensive criminal response upon request. NYSIIS will be notified

of the admission procedure via a fingerprint input from the correction facility and will update its records accordingly.

3. <u>RELEASE</u> (table 4-1 coordinates V-1) Upon release from an institution, NYSIIS will be so notified and will update the individual's record to reflect his new status.

-67-

III. ASSIGNMENT OF PRIORITIES

In order to provide for an orderly and efficient transition to a revised reporting system, NYSIIS will undertake the task in accordance with the following priorities:

- A. NYSIIS is currently engaged in determing the most expedient and advantageous method of collecting the additional data elements which were designated necessary to a parole/probation reporting system by the surveyed agencies.
- B. NYSIIS is currently analyzing those areas of the present reporting system which must be re-designed or modified so as to permit NYSIIS the flexibility required to fulfill the needs of the criminal justice system.
- C. NYSIIS will shortly begin on the analysis and design of a preliminary parole/probation statistical module which will facilitate basic program measurement and evaluation.
- D. NYSIIS plans to initiate discussion with the newly organized New York State Department of Correctional Services (Corrections and Parole) so that NYSIIS may avail itself fully of the resources of this new Department in regard to kinds and transmission of data. Similar discussions will be pursued with the

newly organ 5. NYSIIS plan agencies th possess) co investigate network, an mutual conc

 \mathbf{O}

()

0

0

O

 \bigcirc

 \bigcirc

 \mathbf{O}

D

 C^{*}

port

Ç.,

t

đ

i Ca

5

O

C

С

C

 \mathbf{C}

C

2

C

C

ty,

ion:

nized Division of Probation.	
ins to initiate discussions with the various justice	hibit
throughout the state who now possess (or will shortly	G-2
computerized operations. These discussions will	
te the potentials for an interfacing computerized	
and will seek to identify and define areas of	J-2
ncern and assistance.	

-68-

Upc arı

N-2

0-2

P-2

R-2

S-2

		· · · · · · · · · · · · · · · · · · ·	
c ubit	-69- Legend Table 4-1	ti	pon receipt of an a he individual is cu r if currently on b
tly G-2	Chart indicates that data will be received from court, when in reality data will be received from Judicial Conference of State of New York.	a : a	bove, NYSIIS will n gency and the arres
J-2 () N-2	-do- Dual entry reflects request for assistance information/automatic	r	PL 390.60 requires report and supportin f the sentence is a
0-2 C	violation response. Dual entry reflects request for assistance information/or file registry.		less than 90 days th supply the report.
P-2	In normal process no input at this point since need has been satisfied at M-1. However, cases may arise when information is required.	0	
R-2	Dual entry reflects request for assistance information/automatic violation responses.	,C	
S-2	Dual entry reflects request for assistance information/or file registry.		
	n receipt of arrest notification NYSIIS will always reply to the est agency with an abbreviated history and in addition will reply: - To police (sheriffs) office in rural areas where lower court is	0	
0	of town - village type to satisfy C-1; G-1; K-1. - In urban areas - to police agency to satisfy C-1 and to the arraignment court to satisfy G-1; K-1.		
0		\mathbf{O}	

In arrest notification NYSIIS will ascertain if currently in a parole, probation, wanted status in bail or recognizance, if yes to any of the l notify both the supervising or wanting rest agency.

res that the court forward the pre-sentence rting documentation to the correctional facility is greater than 90 days. When the sentence is there is no such requirement and NYSIIS will

-70-

SECTION V DATA ELEMENT REQUIREMENTS OF THE PAROLE/PROBATION REPORTING SYSTEM

-71-

One of the major areas of investigation undertaken by the survey team, was the identification of those data elements which, in the opinions of the surveyed agencies, were vital to a parole/probation reporting system. The results indicated that much of the necessary information was already being collected from various sources by the present NYSIIS system; however, several other elements were mentioned that were not being collected. The accompanying table (table 5-1) indicates both the data elements presently collected by NYSIIS and included in its data base and those additives deemed necessary to this data base.

Data Element Origin

10

Ċ

C

C

ſ

C

(...

()

 \mathbf{O}

C

C

With very few exceptions, the elements in the current data base and those indicated as additional elements may be collected by NYSIIS from multiple input sources. For example, Birth Place and Birth Date may be submitted as record elements from the arrest agency, the detention and incarceration agency, the probation agency and the parole agency. In a similar fashion, Crime Place and Crime Date may also be reported by a number of justice agencies. In processing items such as these, entries such as birth date/place are defined as identification information and crime date/place are defined as legal information. Generally speaking, once this distinction is made, the processing system proceeds in the following way.

- In processing identification type information, the system incorporates a "frequency" counter which will indicate, numerically, the number of ways a particular individual was reported. For example, using birth date/place, the system will list all the variations on this entry, each accompanied by a numeric indication of the number of reporting instances. This indication is carried on the criminal history report in the "other information" section.

- With legal type information, the system provides for the acceptance of certain data elements by type of event and does not record a frequency. For example, as indicated the crime place/date elements may be received from a number of sources, but it is collected and recorded only at the arrest level, regardless of the fact that it may also be present on disposition, probation, incarceration, etc., submissions to NYSIIS. Certain other legal data items such as incarceration facility will be noted each time they are reported. Thus the incarceration facility, if reported will be noted in the disposition record, the institution admission record, and the institution release record. The above procedures were designated to permit MYSIIS the flexibility of recording all pertinent data on an individual and to present this information to the user in a concise and meaningful manner.

Data Element Usage

()

65

f

11

()

C1

 \mathbf{O}

 \bigcirc

The data elements, once collected by NYSIIS, will be used to satisfy the various information needs of the agencies, and will be tailored to the operating requests and requirements of these agencies. As table 5-1 indicates, most of the data

-72-

submitted to and collected by NYSIIS is that usually comprising data of the "public record" type, and as such is not subject to severe dissemination restrictions.

-73-

Additional Data Elements

As a result of the survey, NYSIIS has been able to identify a number of parole/probation data elements which would be of use to the justice system if collected and reported in a standardized manner. These elements are listed in the "Additional Data Elements" section of Table 5-1, along with the most appropriate source agency and an indication of the potential users of this information. In addition to these elements, the parole and probation agencies surveyed indicated a need for certain data items which were already routinely collected and maintained by NYSIIS, but not widely distributed in the probation area, due to lack of dissemination facilities and procedures. NYSIIS, as indicated in section IV of this report, is taking steps to remedy this situation so as to provide the facilities for comprehensive agency - NYSIIS data exchange.

Crime History Charges File

NYSIIS FILE TYPE

Base File

 \odot

 f_{i}^{∞}

CA

01

Er

 \mathbf{O}

 \bigcirc

0

Disposition File

Institution File

Name File

;ion C

18

ſ

8

° (* -

C -

Ç.,

C

ζ.

C

to

C

and

C

TABLE 5-1

Page 1 of 5

	·	
DATA ELEMENT	SOURCE	TO SATISFY
NYSIIS Number FBI Number Sex Height Race	Various -do- -do- -do- -do-	All Agencies -do- -do- -do-
Skin Tone	-do-	-do- -do-
Arrest Date Crime Date Arrest Place Crime Place Arrest Agency Arrest Number Court Arraignment	Arrest Agency -do- -do- -do- -do- -do- -do-	Court-Correction All Police-Probation-Parole All Police-Probation- Correction Police-Probation-Parole
Arrest Charge Disposition Charge	Arrest Agency Judicial Con- ference/Police Agency	Police-Probation-Parole All All
Charge Literal Disposition Status	-do- -do-	All All
Date of Disposition Court of Disposi- tion	Judicial Con- ference/Arrest Agency -do-	All -do-
Docket Number Restitution Amount	-do- -do-	-do- D.ACourt-Probation-
Fine Type Disposition	do -do-	Parole -do- All
Institution Name Inmate ID Number	Institution Facility -do-	All Probation-Parole-
Date of Detention Date of Admission Type of Admission	-do- -do- -do-	Correction All -do- Probation-Parole-
Type of Release	-do- -do-	Correction All -do-
Nickname	-do-	All Police-Probation-Parole All
-74-		

PRESENT DATA ELEMENTS

TABLE 5-1

E

1-

fe

le le

C

ζr.

е

С

C

Page 2 of 5

NYSIIS FILE TYPE	DATA ELEMENT	SOURCE	TO SATISFY		- (3	NYSIIS FILE TYPE	DATA ELEMENT	SOURCE	TO SATISFY
Mental Hygiene File		M.H. Institu- tion	All Agencies					Inmate NYSIIS Number Inmate Name	-do- -do-	-do- -do-
·	Date of Admission	-do-	-do-			3	License Application	Data of Application	77	÷
		-do-	-do-	10 A	·		File	Date of Application	Various Licensing	Various
		-do-	-do-						Agencies	
	Number				-			Application Number		Depending On
	Date of Demolo	Division of	All					License Agency		Type
Parole File		Parole	A					License Type	-do-	Type License
4	Identification	-do-	Probation-Parole-			- t				
	Number	~~	Police		· · · · · · · · · · · · · · · · · · ·		Employment File	Date of Employment	Various	All
	Supervising Agency	-do-	Parole	-	-			Name of Employer	-do-	-do-
	Parole Length	-do-	Police-Probation-				-		-do-	-do-
		1 · · · ·	Parole-Court		·]			Dept. of State ID	Dept. of	Dept. of State
	Parole Release Date	-do-	All					Number	State	
	Assigned Area	-do-	Police-Probation-				Social Security File	Social Security No.	Vanious	Demals D 1
			Parole		10	3	Social Security FILE	DOCTAT DECULICY NO.	Various	Parole-Probation
		Division of	Court-Probation-Parole				Probation/Custody	Date of Disposition	Judicial Con-	۲ ۲۵
		Division of Parole	0001 0-1 1054 01011-1 a1 016				File		ference/	A77
Relief File		-do-	-do-		and the second se				Arrest Agency	
	Permanency Date Certificate Type	-do-	-do-					Status of Probation		-do-
	Issuing Agency	-do-	-do-		:			Type of Probation		-do-
	Revision Agency	-do-	-do-					Length of Probation	-do-	-do-
	Revision Date	-do-	-do-	· .		5		Status of Sentence	-do-	-do-
	Revision Permanency	-do-	-do-							-do-
	Date									-do-
	Revision Type	-do-	-do-					Institution	-do-	-do-
	Revocation Issue	-do-	-do-				Comments File	Additional Comments	Various	Vontour
	Date		-do-		Automa'iy		Commonial I TTC	war of other commentes	various	Various
	Revocation Agency	-do- -do-	-do- -do-			` a	Associates Name File	Name of Associate	Various	A11
	Revocation Effec-	1-u0-				.)		Associate NYSIIS No.		-do-
	tive Date Revocation Type	-do-	-do-				· · ·			-do-
	revocation type		1							
Birth Date/Place File	Date of Birth	Various	All					Various Items	Various	Various
DI DI DAGGALACC LITO	City of Birth	-do-	Probation-Parole				File	Relating to Non F/P		
	State of Birth	-do-	-do-					Information		
	County of Birth	-do-	-do-			2	Wanted Person File	Nonting Agener	Vontinal A	4 4 4
						<u>.</u>	warred Letzon FITE	Wanting Agency	Wanting: Agen-	ጥጥጥ
Address File	Address Date	Various	PD-Probation-Parole					Transmission Date	-do-	
	Address	-do-	do						-do-	
		Tuetitution	D.AProbation-Parole-						-do-	
Institution Visits	Application Date	Institution	Correction						-do-	
File	Tomata TD	-do-						-	-	
	Inmate ID Institution Name	-do-	-do-		C.	3				n na seni para na seni dena seni dena seni dena seni. 1
	Visitor NYSIIS	-do-	-do-				*** In general all typ	es of justice agencie	s are interest	ed in this type
	Number				and the second		information. Howe	ver, the depth and co	ntent will var	y between agency
	Visitor Name	-do-	-do-		in the second		types.			
	Relationship	-do-	-do-							
					. 12					

 \circ

-75-

0

TABLE 5-1

Page 3 of 5

TABLE 5-1

Page 4 of 5

0

O

1	NYSIIS FILE TYPE	DATA ELEMENT	SOURCE	TO SATISFY	-	0	NYSIIS FILE TYPE
		Sex	-do-			2	
		Race	-do-			1	
	• • • • • • • • • • • • • • • • • • •	Skin Tone	-do-				
		Skin Tone	-do-				•
	6	Ethnic Group	-do-		The second se	1.	
1		Date of Birth	-do-			1 vi	
	9.4. • •	Place of Birth					and a set of the
		Height	-do-		-		
	•	Address	-do-			1	
	1	Social Security	-do-				
		Number					
		Operators License	-do-) .	
		Nimber				C	
1 f		Miscellaneous Number	-do-		1. A		
		Type Warrant/Offense	-do-				
		Date Warrant	-do-			1	
		Case Number	-do-				
		Alias	-do-			[t	
		F/P Classification	-do-				
1		License Number	-do-			IC I	
. 1° .		Vehicle ID Number	-do-			102	
	M	Identifying Marks	-do-				
		Verified Date of	Probation	All		31-14-14-14-14-14-14-14-14-14-14-14-14-14	
	Additional Data	Birth	Agency				
	Elements	Verified Place of	-do-	Probation-Parole			
	ι	Birth					
		Supervision Area	-do-	PD-Court-Probation			
		Transferred to Area	-do-	PD-Probation		0	
14		Transferred to Alea	-do-	Probation-Parole-			
		Verified Marital	-40	Court-DA			
		Status	-do-	All			
		Narcotic Status	-do-	-do-			
		Type of Narcotic	-40-				
		Used	-do-	-do-			
		Psychological	-40-		-	0	
. ₹		Deviations	-do-	-do-			
i i		Employment Data	-do-	D.ACourt-Probation-			
		Occupation	-40-	Parole-Correction			
1			-do-	Probation-Parole-		-	
		Education Data	-u 0-	Correction			
			-do-	Probation-Parole			
1		Military Data Bail/ROR Data	-do-	Police-DA-Court		0	
1		Bail/RUK Data	-u0-	Probation-Parole		Q	
· · · · · · · · · · · · · · · · · · ·		Danal Baral	-do-	D.ACourt-Probation-		1.	
		Attitude on Prob	1-00-	Parole			
		Discharge		Probation-Parole			
1		Type Investigation	-do-	110040100-141010			
	92-9 1974 1975	Performed		-do-			
		Availability	-do-	-do-			
		Location	-do-	PD-Court-Probation-		10	
	- 6	Supervising Agency	-do-	Parole			
				Parole Probation-Parole			
		Probationer ID	-do-	Police-Court-Probation-			
and the		Transferred From	-do-				
				Parole			
						lio	
€		-77	-			² د ب	
			l i j				
10						I	

C

TABLE 5-1

Page 5 of 5

_	the second se			s s
-	DATA ELEMENT	SOURCE	TO SATISFY	
	Type Supervision Cooperative Supvsn.	Probation/ Parole	D.ACourt-Probation- Parole	of
	Data -Original ID No. -Original State	-do- -do-	Probation-Parole D.ACourt-Probation- Parole	pro pre
	-New Supervising State -New ID Number	-do- -do-	-do- Probation-Parole	of
-				vi]

I.

SECTION VI

-79-

PAROLE/PROBATION REPORTING SYSTEM

In order that NYSIIS may accomplish the stated survey objectives of satisfying the needs of criminal justice agencies for parole/ probation information and provide for the rendering of the services previously described in this report, NYSIIS will undertake a series of revisions to its primary reporting system. These revisions will encompass the following points.

- 1. revision of output format to more closely provide the critical information needed by processing agencies throughout their operations;
- 2. revision of dissemination procedures so as to facilitate the transmission of this information in a time frame acceptable to agency needs:
- 3. revision of current data collection procedures to permit NYSIIS to acquire more comprehensive information concerning individuals of interest to the system in a more expeditious manner;
- 4. revision of the basic composite of the criminal history reporting system to provide for, in addition to a data base file, a status file and a statistical file.

The report thus far has examined the manner in which NYSIIS will satisfy points one and two above. In this section, the report will deal with the procedures necessary for the implementation of points three and four.

I. REVISION OF CURRENT DATA COLLECTION PROCEDURES

1

ļľ,

1

1

Ei

 $\{C\}$

As the section on required data elements indicated most of the data required (including parole/probation data) is presently being collected. However, in regard to those elements not presently being collected, NYSIIS will expand its data collection procedures by requesting more active input assistance from the probation and parole agencies. One of the

areas of major concern is the receipt, from probation, of notification of prior investigation reports so that NYSIIS may include this information on its responses with the objective of reducing the investigative workload. Another area of concern for the system is the acquisition of timely results of bail/ROR proceedings so that the system may be continually aware of the status of an offender complete through the process. The collection of this data causes problems which are quite similar to those problems mentioned in section IV of this report dealing with the dissemination of data. In order that information such as the above, particularly bail/ROR information, be reported to and processed by NYSIIS within an acceptable time frame, this info mation must be quickly transmitted to this agency. This transmission requirement cannot, in most instances be satisfied through normal mail deliveries, hence NYSIIS plans to utilize much the same procedure as previously described for output dissemination, for receiving input da That is, in urban areas NYSIIS would request, that, via centralized court facilities, the probation agencies input the preliminary disposition data to NYSIIS for immediate update of the central files. In the rural areas, NYSIIS plans to request the assistance of the agency usually State Police or Sheriff's office, possessing the transmission equipment to forward the necessary data. II. STRUCTURE AND PURPOSE OF THE COMPUTER SYSTEM

As indicated in the previous NYSIIS survey report, dated April 1970, the present NYSIIS computer criminal history reporting system consists of over twenty sub-files making up the NYSIIS data base file These files contain the total information available on an individual and it is via these files that responses are produced for agency use.

C

C

14

 $\langle \cdot \rangle$

C

|

1

10

()

 $\hat{\Box}$

- 80-

₹⁸. Σ ٢ this O invehe plete s whic his formaporte infor ssion l mail e as ut dat zed ispo-(CI In F gency ssion G = 1ril cem filé O Ldual use O

system.

A. STATUS FILE

In order to streamline the above procedure, NYSIIS plans to introduce a status file into its present criminal history reporting system. This file will contain pertinent extracts, from an individual's complete record, which will reflect the current criminal justice status of the individual. These extracts will consist of entries which will chronicle the individual's progress through the system from point of entry (arrest) to point of exiting the jurisdiction of the system (release from legal obligation). At a future date the status file organization will provide for the entry of post-sentence "appeal" events, however, at the present time due to the absence of comprehensive data concerning this area it has not been included. The following narrative describes what is anticipated to be representative of the typical operations of the status file (see chart 6-1).

At the present time, although the data exists, the system does not possess the capability for the production of statistical data or for an expeditious method of determining the present criminal justice system status of an individual. Presently in order to determine this status, NYSIIS must scan the individual's complete file, and produce a comprehensive history report for transmission to the input or inquiring agency. Based on the data in the report the recipient agency then determines the individual's status within the

-81-



~____

1. ARREST NOTIFICATION

C.

C

ſ

C

C:

10

 \mathbf{C}

Upon receipt of an arrest fingerprint record and following the manual and computer data verification process, an update concerning this event will be performed on the status file to indicate entry into the justice system.

-83-

2. PRELIMINARY JUDICIAL RESULTS

Upon receipt of preliminary data, two processing alternatives are possible. If the data represents a dismissal of charges the system will provide for the purging of the arrest information from the status file, since the individual is no longer a matter of system responsibility. However, should the charges remain, the offender may be admitted to bail or recognizance release or be remanded to a detention facility to await further judicial action. When any of these occur, the individual's status file will be updated to reflect this condition.

3. PRE-TRIAL JUDICIAL RESULTS

This stage of the judicial process generally consists of the activities following a grand jury indictment. Again, as in the previous instance, the individual may qualify for bail or recognizance release or be remanded to custody. The entry at this point will reflect this judicial decision.

4. SENTENCING RESULTS

If, as a result of the trial process, the defendant is either acquitted of all charges or sentenced to a period of imprisonment the information relating to

£

C

6

Ô

0

this event will be purged from the status file. This purge will be performed, in the case of acquittal, for the same reason as the purge process at the charge dismissal point. In the case of a sentence of imprison-

ment, the purge will be performed due to the fact that the individual may no longer be considered in an active status. If the offender is convicted and sentenced to a period of probation, this information will be entered onto the status file.

5. RELEASE TO PAROLE SUPERVISION

Upon receipt of notification informing NYSIIS of a parole supervision status, this information will be posted to the status file.

в.

6. RELEASE FROM JUDICIAL OBLIGATION

Upon an individual's successful completion of a probation, parole or conditional discharge supervision period, all information relating to the original event will be purged from the status file. If, during the supervisory period, the individual is returned or is sent to a correctional facility, all records pertaining to the original event will be purged from the status file.

7. ADVANTAGES OF THE STATUS FILE

The implementation of the status file will provide all criminal justice agencies with the rapid type of notification they require regarding the status of parolees and probationers. It will provide for the identification of new arrestees who are currently in a phase of either postadjudicatory (parole-probation) supervision or pre-adjudicatory (bail-recognizance) supervision. An added benefit to the system through the utilization of this file will be recognized in the assistance it will lend to all those agencies performing investigative operations. For example, in the police crime investigation phase, this file is seen as potentially reducing the number of suspects the agency must concern itself with. In another way this file is seen as lending assistance to parole and probation investigative officers in their offender or offender-associate types of investigations by providing rapid notification of the subjects current justice system status.

-85-

STATISTICAL FILE ₿ **В**.

¢

on-

 $t_{\mathcal{C}}$

٧e

eđ

C

С

C

C

C

1

f**§-**

tion

•

7 9

8

The second planned additive to the current criminal history reporting system will be a statistical file. The addition of a statistical capability will provide NYSIIS with the operational potential for providing either comprehensive statistical reports for individual agency evaluation or for providing those data elements which are required by the individual agencies in performing their own evaluation of programs.

The statistical file will contain those data elements concerning offenders which will enable parole and probation agencies to effectively evaluate the effects their operations and programs and the total justice system have had on the individual passing through the system. In order for this evaluation to be complete, it must collect data at system entry, at various points throughout the system and at system

exit. The input to the system would initially consist of presently received documents plus those additional elements relating to the interim judicial process. At a future date, once experience has been gained, the system will provide for the acceptance of additional data elements which will assist in providing the mechanisms for more detailed statistical analysis. It is anticipated that these additional elements will consist of information relative to types of institutional programs and training enrolled in and the results of such training; similar information relating to planned probation programs and training, etc. This data would be captured and correlated on the file in a manner which will optimize its utilization by the system's subscribers.

III.

The survey results indicated that, as a whole, the supervisory agencies are actively interested in receiving data and assistance from a central agency which will facilitate their operational evaluative process. Of particular concern to many of these agencies was the acquisition of subsequent information concerning individuals who had previously been under their supervision and discharged. It was indicated that by examining these subsequent criminal activities or lack thereof. (in addition to previous criminal behaviour) an agency would possess the basic ingredients necessary to initiate research into the effectiveness of present agency procedures. The output of the planned statistical system would be structured, as far as possible, in accordance with basic user agency goals and automated capabilities. Thus, the output of the system is seen as providing either individual or broadly based (by

-86-

2

3

¢,

٢

8

٢

۲

10

٢

0

pre-arranged classification) statistical reports or, the basic data necessary for an agency to produce its own reports. The following section, Section VII will describe in greater detail the data needs and potential uses of this file in providing the sound basic statistics necessary for effective program planning.

-87-

OPERATION OF THE SYSTEM

1

br

;t

€

)nal C

n.

'nđ

С

)Cr-

and

1.r

neny

ation

LC-

(in

se

вd,

orls

C

II.

The manner in which NYSIIS will process the documents it receives will be divided, for discussion purposes, into two segments, namely the INQUIRY PROCESS segment and the UPDATE PROCESS segment. The following is a brief explanation of the major functions of each process in relation to its interaction with the NYSIIS computer based files and the type(s) of reports generated by each process segment.

A. INQUIRY PROCESS SEGMENT

This process (see chart 6-2) is designed to handle those types of input documents which only request information from the files. In determining the manner in which these requests will be accommodated, the system will provide for alternate processing procedures depending upon the type of request.

In those cases where the inquiry is a request for statistical data which will assist in the research and statistical operations of an inquiring agency, NYSIIS will search the statistical file; access the data; and release it in the format required for agency use. The other possibility in inquiry processing includes requests for prior history information by justice system agencies. In processing these



requests, NYSIIS will search both the status file and the data base file in order to produce one of the types of responses described in Section IV of this report.

B. UPDATE PROCESS SEGMENT

C

C

C

C

C

C

C

€

C

C

This segment (see chart 6-3) provides the mechanisms whereby the three primary NYSIIS files (data base; statistical and status) are updated with current information concerning individuals in the justice system. In general, the operations of the segment may be described as follows:

- Upon the receipt of a criminal justice event, the system will provide for the immediate addition of all pertinent data to the statistical file. The type of data entered on the statistical file will be discussed at greater length in Section VII of this report. At this point the process segment will decide whether or not the event is an arrest event and accordingly will proceed in one of the following ways:

- if the event is an arrest event and is also the first event for the individual, the system will initiate records for the individual on both the data base and the status file and will format a reply for transmission to appropriate agencies.
- 2. if the event is an arrest event and is a subsequent event, the system will update the data base. Additionally, if the offender is recognized to be in an active status, via the status file, the system

in the However active in the appropriation if the represe system individ if the represe type (hupdate

)

cal

•

:in

15

۲

£

٢

-88-

will update the status file and notify both the arrest agency and the agency, exercising interest in the individual as indicated by the status file. However, if the offender is not found to be in an active status, the system will initiate a record in the status file and format a reply for the appropriate agencies.

3. if the event is other than an arrest event and represents a removal from active system status, the system will purge the status file and update the individual data base file.

¹4. if the event is other than an arrest event and represents additional system data of an active type (bail-ROR-probation-parole) the process will update the status file and the data base.







SECTION VII

CRIMINAL JUSTICE STATISTICAL REPORTING SYSTEM

In order that the statistical file concept discussed in the previous section be utilized to its utmost not only by parole/probation authorities but also by all segments of the justice system, NYSIIS is planning to develop and implement a series of procedures which will provide for an eventual comprehensive criminal justice statistical system.

The initial stage of this statistical reporting system will require that NYSIIS automatically collect the type of data now included in the SEARCH statistical project demonstration system. In defining the data elements to be utilized, in this initial stage, NYSIIS will closely conform to the recommended items contained in the Data Collection Manual for the SEARCH Statistical Demonstration Project, dated March, 1970. Most of this required data is routinely collected by NYSIIS in its data base updating process, and when required it will also be included on the statistical file. The acquisition of certain other required data (e.g. iterim judicial determinations), although not presently provided in all cases, via the survey, been recognized as necessary to a full information system, and procedures are currently being developed to provide for the timely receipt of this information by NYSIIS. Certain other information required by this system (e.g. presence or absence of a pre-sentence report) has never been routinely received by NYSIIS, but the facilities are available for doing so in the future.

(Effective September 1, 1971, a pre-sentence investigation will be mandated by CPL 390.20 for all felony convictions and for all misdemeanor convictions unless the sentence consists of a fine or a period of incarceration less than ninety days duration). With the implementation of this stage of the ongoing automated statistical system, NYSIIS will possess the capability of satisfying, on a higher volume and more immediate scale, the original objectives of this statistical system as established by the SEARCH Statistical Methods Task Force. This statistical system will trace offenders, on an individual basis, through the New York State criminal justice process and, as in the national system, will be designed to indicate by major system decision point the results of system process on an individual.

The successful completion of this preliminary stage of a state-wide justice system statistical model will precipitate the advancement of the model to a higher developmental stage. A major objective of this more advanced stage will be to provide statistical information on an individual transaction basis for a representative sample of individual justice system agencies. The type of data required to achieve this objective is basically similar to the previously discussed elements of data; in addition, this phase will require the collection and utilization of specific agency identities. A second objective of this stage will be to examine the feasibility of producing statistics such as: -Individual transaction statistics reflecting the flow of the system on similar type offenders (e.g. same age-racesex-charge, etc.) on a state-wide level;

-92-

(

(

Č.

-93-

 \bigcirc

18

()

6,

()

()

IC:

63

()

 \cap

G

-Individual transaction statistics reflecting the flow of the system on similar type offenders within a specific agency designation (e.g. statistics reflecting the system flow on all 25 year old males arrested for robbery, who were or are being supervised by a specific probation agency);

-94-

C

C

· ()

£ 3.

(a

C

-Cumulative individual statistics which will reflect, on the state level, the system flow on individual offenders with multiple events;

-Cumulative individual statistics which will reflect, via a specific agency designation, the system flow on individual offenders with multiple events.

The basic data components suggested by the SEARCH statistical task force are with the exception of specific agency designation sufficient to accomplish the above objectives.

Once the above stages have been completed, modifications or refinements implemented and the subsequent results analyzed, the project will progress to a third and more comprehensive phase. This phase will include all the attributes of the previous stages and will contain basically the same information. However, it will be designed to include elements which will greatly refine the overall results, such as educational, marital and military experience. They will encompass items concerning employment and training experience (institution or private). In short, this phase of the statistical system will incorporate all those elements which the surveyed agencies indicated were vital to a comprehensive and accurate analysis of agency programs, policies, procedures and results. This phase will provide statistics which will not only reflect the flow of the system via its various "drop-out" points, but will also reflect the types of background commonality among offenders. In addition, this phase may be extended to provide the detailed data necessary for the examination and appraisal of both the criminal justice system and its individual component agencies as positive reconstructive forces.

Data Requirements of the System

Es.

(s

 \int

 \bigcirc

6

G

 \bigcirc

O

The initial portions of this section alluded to the type of data required to produce a preliminary, intermediate and comprehensive version of a statistical reporting system. Admittedly, these elements exist in various formats in many segments of the criminal justice system and are used for various purposes (see table 5-1). So, in order to provide for the systematic collection of this data; to avoid the redundant entry of similar data to the statistical system, and to demonstrate the practicality of data accumulation and control at a central function, NYSIIS has defined the type of data it will collect and maintain from each type of agency.

The following data breakdown is not intended to be totally inclusive, since system experience will, in all probability, necessitate modifications to some degree.

+95₩

1. Arrest Agencies

Data is needed concerning frequencies of reported and (where possible) unreported crime; number of arrests; types of arrests (i.e. warrant, citizen, etc.); type of arrest offense; type of booking offense and complete information concerning the pertinent information surrounding the arrest (i.e. dates, places, etc.).

-96-

2. Judicial Agencies

C)

C

С

C

At the pre-trial stage (preliminary arraignment/initial court appearance, etc.) data should be available in relation to bail, (including amount), the number of persons released on their own recognizance (ROR programs), the number of persons detained prior to trial, reasons for and length of their detention, the number of persons not appearing for trial and the reasons for non-appearance.

At subsequent judicial levels, data should cover adjudications; number and type of trials; number of cases in which there was no trial; sentences and lengths; whether or not the accused was represented by counsel, and whether counsel was retained or appointed; number of postponements or continuances and the reasons for these. Finally, it should be recognized that data concerning the time expended in court proceedings must be available and should, in addition to indicating total judicial time for a complete proceeding, be refined to permit the relation of time spent in and between the major phases of the court process.

For example: a. decision)?; b. hearing and trial?; d. How much time was expended in the actual trial phase? 3. Probation Agencies 4. Parole Agencies reasons for these infractions.

(3

63

 \bigcirc

Fi

63

C

O

G

6

8

B

How much time was expended from point of arrest to ultimate final disposition (Sentencing or appellate

How much time was expended between conviction or guilty plea (trial) and sentence date?;

c. How much time was expended between the preliminary

At the probation level, data is needed on type, availability, and location of various investigations and their reports; number of persons on probation; type of specific probation program enrolled in; social background data; verified identification data elements (e.g. date of birth, true name, actual residence and place of employment, etc.); and specific data relating to probation revocation activities such as cause of revocation (i.e. technical violation and type, new arrest, etc.); number of hearings, etc.

NYSIIS would require that Parole furnish data concerning types of parole programs parolees are enrolled in; the number of parole board hearings; the number of paroles approved; the relation of parole violations to the three general categories of new offense, technical violation and absconding, and the

-97-

5. Correctional Facilities

C

C

C

C

Ĉ

Data needs include the number of persons in correctional institutions (local and/or state); the reasons for incarceration; whether or not these people are awaiting trial or are sentenced inmates; their sentences; number of persons discharged and released on parole; numbers being trained in educational and vocational programs; the types of programs involved; and the numbers in pre-release programs.

-98-

In essence, what is needed is a uniform statistical reporting system to serve all criminal justice agencies by providing information which will assist them in assessment of policies and programs and in planning more effective crime prevention, treatment and control measures. Before much progress may be made in examining the interrelated decisions along the criminal justice process from arrest of offenders to their final discharge, we need reliable comparable information about the persons involved, the decisions made about them, and the consequences of these decisions. Once this data is obtained and correlated, it will be possible to begin examining the results of decisions in one sector of the criminal justice system, upon other criminal justice sectors. The police, probation officers, judges, correctional workers, parole boards, and others involved in the administration of criminal justice all have at least one thing in common -- they all must make decisions concerning criminal offenders. Yet these decisions, each of which is an integral part of the total system, are often unrelated to their effects on the

entire system. The statistical system discussed here will assist in the total relation of the operations of the system.

6

()

64

 \bigcirc

 \mathcal{O}

()

 \mathbf{n}

0

(Cr

()

ATTACHMENT A

-100-

Proposed Probation Data Elements

1. ADMISSION TO PROBATION NAME (LAST, FIRST, M.I.) NYSIIS NUMBER (SID) SUPERVISION AREA DATE RECEIVED AT SUPERVISION OFFICE VERIFIED DATE OF BIRTH LENGTH OF PROBATION TYPE INVESTIGATION PERFORMED AVAILABILITY OF INVESTIGATION LOCATION OF REPORT LOCAL IDENTIFICATION NUMBER

2. VIOLATION

6

0

1

¥".

1

|C|

C

C-

C

C

NAME (as above) NYSIIS NUMBER (SID) RESULTS OF PROCEEDINGS WARRANT INFORMATION

3.

TRANSFER NAME (as above) NYSIIS NUMBER (SID) NEW SUPERVISING AGENCY DATE OF TRANSFER NEW LOCAL IDENTIFICATION NUMBER

4. RELEASE

NAME (as above) NYSIIS NUMBER (SID) DATE OF RELEASE PROBATION PROGRESS

			* · · · · · · · · · · · · · · · · · · ·	101	ATTACHMENT B	-1 PAGE 1 OF 2	
				- 101 - (- EXECUTIVE DEPARTMENT	ALLAOIMIDML D	THUS I ON E	
			IDENTIFICATION AN	ID INTELLIGENCE SYS	TDANK	• · · · · · · · · · · · · · · · · · · ·	
- 1	DATE 6=03=70		ALFRED E. SMITH STATE OFFIC	CE BLDG., ALBANY, NEW YO	RK 12225 NO.		ATE C
	NYSIIS 0078301A	THIE DEED	ONSE TO YOUR IN	TRY IS BASE	O ON A F/P II	FNT	NO.
	NQUIRY SAMUEL SA				ALE HGT 6=0		QUIRY UAME NOWN
. *	AS SAMUEL SA	HPLE				No.183-32-1749	AS
		· •					
-			UREAU OF INVEST	LOATION			
	· · · · · ·	FEDERAL B					
· _	· · · · · · · · · · · · · · · · · · ·	ALBANY N	.Y. 12207				66
~		······		· · · · · · · · · · · · · · · · · · ·	····		03.
		•	CRIMINAL	HISTORY			
-							
	ARRESTI ADDRESSI	SAMUEL 121 3RD	SAMPLE	NEW YORK		ICY ID B1846672 I York	
_	GRIME DATE		01=31=60	NEW YORK		YORK	
	ARREST DAT		03-02-60	NEW YORK	7.5.7.7 State 12.1	YORK	NĂI
		53RD PRECINC		NEW YORK	CITY NEW	YORK	FF
~	-ASSOCIATES			0600		ICY ID 8024956	RE (
		SAMUEL			AGEN	CY ID 8024956	03
	ADDEET OUA		BERT JONES SSION DANGEROUS	DDUA	AGEN	ICY ID 8124956	
_			ING DANGEROUS WI				
	DISPOSITION		04=31=60		DOCK	ET NO 00003333	
-			ININAL COURT	NEW YORK		YORK	<u>{</u>
			ANGEROUS DRUG				- 1
			GEROUS WEAPON 18				1
-	SENTENCEI	5 YEARS	NVICTED PLEA OF	WILI -	GL	IMBINED CHARGES	
		SING SING ST	ATE PRISON	OSSINING	NE	YORK	
_		ADHISSION:	05=05=60			TE 10 01234567	
		SING SING ST		DSSINING	NE	YORK	1
			NOT ON PAROLE			TE 10 01234567	s b ara
·	INSTITUTION	SING SING ST	05=04=62	USSINING		I YORK	
	PAROLED	31 MU 31 MU 31	HIE FRIGUR	UDJININU	NE I		
_	PAROLEI O	5=05=62 3 Y	EARS		AGEN	CY 10 0078301A	
		DIVISION OF	PAROLE			YORK	. 1
	CERTIFICAT	E OF RELIEF.		E DATE 06-12-			
-		DIVISION OF		ALBANY	NEW		
		RELIEVE HOL	ICATE SHALLI DER OF ALL DISA	BILITIES AND	BARS TO EMPLO	IYHENT,	1
			IGHT TO BE ELIG				
	ARRESTI		SANPLE			ICY ID 00276538	:] .
-	-ADDRESSI	5366 15	02=26=68	NEW YORK		L YORK	
	ARREST DATE		02=26=68	NEW YORK Leeds		YORK	1
-	AGENCY	N.Y.S. POLI		LEEDS		YORK	
	ARREST CHA	RGESI PL 1	20,05-01 CLASS	FEL ASSAUL	T 2ND		
	DETENTION		02-26-68			TE ID 00041736	
		GREENE COUN		CATSKILL-		TT NO DODDOD	
	DISPOSITION	I Manhattan Cr	03004=68			(ET NO 00000001 I YORK	
÷.,			I CLASS A HISD	ASSAULT 3RD			
						9) SUMMARY CASE HISTORY	
	* Represente manat 1-f	mation unsupported by fing		Ω.	J R. J. Fallat		* Repr
	vehierents diter 1010	mation unsupported by ting s complete as the data furn	erprints in our tiles.	· · · · · · · · · · · · · · · · · · ·	111. V. Alla		

0

0

10

-

				1 D	
-101- ATTACHMENT B-	1 PAGE 1 OF 2			2	
STATE OF NEW YORK- EXECUTIVE DEPARTMENT					
IDENTIFICATION AND INTELLIGENCE SYSTEM ALFRED E, SMITH STATE OFFICE BLDG., ALBANY, NEW YORK 12225 NO.	12857 PAGE 1	ATE 6			ATE 6-03-70
		- 11			
NYSIIS OO78301A THIS RESPONSE TO YOUR INQUIRY IS BASED ON A F/P IDE NOURY SAMUEL SAMPLE TO NOURY IS BASED ON A F/P IDE TONNEDIUM SEX MALE HGT 6-01		NO. O(NO. 0078301A THIS RES
NOULIRY SAMUEL SAMPLE TONMEDIUM SEX MALE HGT 6-01		NOWN SA			HAME SAMUEL SAMPLE
AND SAMUEL SAMPLE APP.CAUCASUID BIRTH U7-16-31 SEC. NO	0.103-36-1147	AS S /			AS SAHUEL SAMPLE
				1	
TO: FEDERAL BUREAU OF INVESTIGATION		ACT			ACTION: 03-04-68
FEDERAL BUILDING Albany: N.Y. 12207	· · · · · · · · · · · · · · · · · · ·	SEN		1.	SENTENCEI - 90 DAYS-
					•
					• •
		03-2			03-28-66 SAMUEL SAM
CRIMINAL HISTORY					STAMFURD CI
	CY ID B1846672 York			\$	
	YORK				
ARREST DATE/PLACES 03=02=60 NEW YORK CITY NEW	YORK	NAHE			NAMES USED BY SUBJECT
AGENCY: 53RD PRECINCT NEW YORK CITY NEW	YORK	FRE Rego			FREU 04 SAMUEL SAMPLI
	<u>cy ID 8024956</u> Cy ID 8024956	03-			03-02-60 121 3
	CY ID B124956	02-		E.	02-26-68 5366
ARREST CHARGES - POSSESSION DANGEROUS DRUG		BIRT			FREQ 04 07=16=31 FR
CARRYING DANGERDUS WEAPON 1896 PL DISPOSITION: 04-31-60 DOCKE	ET NO 00003333	FAL			FUEL OF TELEVILLE FR
	YORK				
CHARGES: POSSESSION DANGEROUS DRUG					
CARRYING DANGEROUS WEAPON 1896 PL Action: 05=04=60 convicted plea of guilty con	BINED CHARGES			a	i.
ACTION: 05=04=60 CONVICTED PLEA OF GUILTY COM Sentence: 5 years					
SING SING STATE PRISON OSSINING NEW	YORK				
	<u>TE 10 01234567</u> York				
NEW COURT COMMITMENT, NOT ON PAROLE					
INSTITUTION RELEASE: 05-04-62 INHAT	TE 10 01234567			0	-
	YORK				
PAROLED AGENC	CY 10 0078301A			Į	
DIVISION OF PAROLE NEW	YORK				
CERTIFICATE OF RELIEFORIGINALI ISSUE DATE 06-12-65	nor		1		
THIS CERTIFICATE SHALLS					
RELIEVE HOLDER OF ALL DISABILITIES AND BARS TO EMPLOY	THENT			0	
EXCLUDING RIGHT TO BE ELIGIBLE FOR PUBLIC OFFICE					
ARREST: SAMUEL SAMPLE AGENC	CY ID 00276538				
ADDRESSI 5366 1ST AVE NEW YORK CITY NEW	YORK				
	YORK York				6
	YORK			<u>[]</u>	
ARREST CHARGES: PL 120,05-01 CLASS D FEL ASSAULT 2ND					
	TE ID 00041736			}	
	ET NO 00000001				* executive
MANHATTAN CRIMINAL COURT NEW	YORK				
CHARGESI PL 120,00=01 CLASS & RISD ASSAULT 3RD NYSIIS-5 (1/69)) SUMMARY CASE HISTORY	 .		-13-	et
β β β					
* Represents arrest information unsupported by fingerprints in our files. All entries are as complete as the data furnished to NYSUS		Represents All			Represents arrest information unsupported by All entries are as complete as the data
			an an an Array		
					1

C

- OTHER C		tv		
	RIMINAL HIS	TORY	•	
Y JAIL For Vagra		FORD	CONNECTICUT	
OTHE	R INFORMATI	ON		· · ·
	FREQ O	1 "LEFTY"		
VE A VE		YORK CITY York City	NEW YORK New York	
		FREQ 04 NEW Y	ORK	· ·
╶╃╴ݠ╶╩╴╇╶┶	AST PAGE -	A. H. A. H		
<u> </u>				
			• ••••••••••••••••••••••••••••••••••••	· · · · · ·
· · · ·	· · · · · · · · · · · · · · · · · · ·			· · · · ·
· ·				
· · · · · · · · · · · · · · · · · · ·				
، مغرب میں معرفی میں میں میں میں میں میں میں میں میں می				

ATTACHMENT B-2 PAGE 1 OF 2 -103-

TRAN NO. 12857 PAGE 1 7=01=70 \odot 7=0 1. 0078301A THIS RESPONSE TO YOUR INQUIRY IS BASED ON A F/P IDENT MALE 6-01 04273738 007 SAMUEL SAMPLE MEDIUM 07=16=31 183-32-1749 SAM SAMUEL SAMPLE CAUCASOID SAM CONFIDENTIAL TO: FEDERAL BUREAU OF INVESTIGATION \odot 1 ТН FEDERAL BUILDING ALBANY, N.Y. 12207 ASS THO CURRENT STATUS SUBJECT CURRENTLY ON PROBATION TU LOUI NASSAU CO. OFFICE OF PROBATION 0 0 ARREST - JUDICIAL INFORMATION ADDE 27 - 2 COURT ARREST CHARGE JUDICL DISPOSITION 260 ARREST ARREST DECISION DATE CHARGE AGENCY DATE EMPI PL 12525 01 A FEL 1 ** 102767 PL 12525 01 A FEL 1 CHARGE 0902 090267 NYPD \bigcirc ſ WITHDRAWN MURDER 1ST 0117 BO17219 MURDER 1ST 1222 011768 NASSU CO PL 22015 02 D FEL 2 ** 031568 PL 22010 00 E FEL 3 PLEA OF CRIM POSS DANG DRUG GUILTY 30172 CRIM POSS DANG DRUG ** EDU HIGH 082569 SCHEN PD PL 21545 00 A MISD 0 ** 110469 PL 21545 00 A MISD 0 ACQUITTED CUMPOUND CRIME \cap MILI CUMPOUND CRIME ** 271 C BRAN 122269 NASSU CO PL 14017 02 U FEL 1 ** 032970 PL 14015 00 A MISD 2 CONVICTED DATE CRIMINAL TRESPASS 30172 CRIMINAL TRESPASS 1 ** PL 14025 01 C FEL 2 NOT ARAIG PL 14025 01 C FEL 2 ** THIS CHRG BURGLARY 1ST ** 050170 BURGLARY 0 042070 NYPD PL 14030 01 B FEL 1 ** \mathcal{C} BO17219 BURGLARY 1ST ** POST-ADJUDICATORY INFORMATION COMMENTS SENTENCE CHARGE ACTIVITY DĂTE \bigcirc PL 22010 00 E FEL 3 1 YEAR ATTICA STATE PRISON C041968 IMPRISUNMENT 101868 REL TO PARULE PL 22010 00 E FEL 3 011669 REL FRUM PAROLE PL 22010 00 E FEL 3 EXPIRATION PL 14015 00 A MISD 2 SUPERVISION PERIOD 18 MOS 040270 PRUBATION NASSAU CO PROB OFFICE MENTAL HYGIENE INFORMATION \square COMMENTS PATIENT DISCHG INSTITUTION NAME-LOCATION ADMIT DATE ID NU DATE 091669 B-072116 100469 BELLEVUE HOSPITAL PRISON WARD PRE TRIAL PSYCHIATRIC EXAM NY NY OC

7=01=70

3

0

THOMAS A WINTERS LOUIS T NOITS

00783014 THIS RESPONSE TU YOUR INQUIRY IS BASED ON A F/P IDENT SAMUEL SAMPLE MEDIUM MALE 6-01 04273738 SAMUEL SAMPLE CAUCASOID 07=16=31 183-32-1749 ITHER HISTORY INFORMATION ASSOCIATES SIU NO. 3015017 SID NO 3015018 ADDRESSES BINTH DATE/PLACE 27-25 QUEENS BLVD JAMAICA NY FREQ 03 QUEENS NY FREQ 05 260 NASSAU DH MINEOLA NY FREQ 02 EMPLOYMENT

090267 UNIVERSAL DIAPER CO BABYLON NY FULDER 011768 UNIVERSAL DIAPER CO BABYLON NY FULDER 122269 DOWNTOWN TRON WORKS WELDER NY NY

EDUCATION HIGHEST GRADE 11TH

MILITARY BRANCH US ARMY DATE OF DISCHARGE 09-15-53

0

0

0

ATTACHMENT B- 2 PAGE 2 OF 2

-104-

TRAN NO. 12857 PAGE 2

SERVICE NO 127305794 TYPE DISHONDRABLE

 \bigcirc -106--105-ATTACHMENT B-3 APP ENDIX MOPSIZED DEFINITION OF LEGISLATION MENTIONED THROUGHOUT THE TEXT \mathbf{O} NOPS TO: COUNTY PROBATION DEPARTMENT ANYTOWN, NEW YORK 12345 W YORK STATE SENATE BILL 9086-A YO this bill, with an effective date of January 1, 1971, amends thi Cboth the Correction and Executive laws of New York State and NYSIIS NO. 0078301A RE: SAMUEL SAMPLE bot provides for the removal of the State Division of Parole from pro the Executive Department and consolidates it with the Department THE ABOVE INDIVIDUAL IS CURRENTLY LISTED AS the of Correction in a newly created Department of Correctional of Services. BY YOUR AGENCY. Ser UNDER SUPERVISION CI WANTED W YORK STATE SENATE BILL 9097 YO 987654 YOUR AGENCY I.D. NO. this bill with an effective date of January 1, 1971, amends both the Correction and Executive law of New York State by providing thi for the removal of the State Division of Probation from the State the SUBJECT IS PRESENTLY IN THE CUSTODY OF \bigcirc for Department of Correction and re-establishes it as a division Depa within the Executive Department. with ANYCITY, NEW YORK P.D. W YORK STATE SENATE BILL 7276 W YORK STATE ASSEMBLY BILL 4561 THEIR AGENCY I.D. NO. 456789 YOE YOE O jointly enacted legislation with an effective date of September 1, 1971, providing for the repeal of the New York State Code of Criminal Procedure and establishing in its place a Criminal Projoir l, 1 Crin cedure Law (CPL). This newly enacted law, contains provisions relative to criminal actions and proceedings throughout the State cedu 0 rela of New York. of N Re following is a brief narrative of the major contents of those e fol ections of the CPL mentioned in this presentation. $|\alpha|$ ction L 160.10 L 160 - fingerprints are required to be taken in the following instances: fin -arrest for felony

0

10

C

 \bigcirc

()

 \mathbb{O}

-arrest for penal law misdemeanor -arrest for misdemeanor defined outside the penal law, which would constitute a felony if the individual has a previous judgment of conviction for a crime. -loitering, as defined in subdivision three of section 240.35 of the penal law.

0

 \square

0

	-107-				
			· · ·	0	
CPL	160.20	CPL			CPL 530.20
	manuface that discommints token in accordance with manufaceuts				E 200 E 20
-	requires that fingerprints taken in accordance with requirements of CPL 160.10 be forwarded to NYSIIS.			S	- directs that,
	OI OI H 100%10 DC IOIWGIGGG 00 MIDIID.				offenses of ie
					recognizance o
CPL	160.30	-			- further direct
					nizance or bai
	requires NYSIIS to classify all fingerprints received, pursuant to CPL 160.20, search its records for pertinent information and			\circ	unless and unt
	transmit such information, or notification of lack thereof, to				m
	the forwarding police officer or agency.				- The dist
					right to
	160 10			0	- The cour
CPL	160.40	ŀ		0	concerni
	upon receipt of a fingerprint report from NYSIIS, the police	1.00			
	officer or agency must promptly transmit such report or a				
	copy thereof to the district attorney of the county.				
				O	
MDT	390.10				
	340.10				
_	directs that the court may not pronounce sentence in the event				
	of a conviction for any offense specified under CPL 160.10				•
	until it has received a fingerprint report from NYSIIS.			Ø	\$
CPL	390.20				
	directs that, if a felony conviction, the court must order a		an an tha an that		
	pre-sentence investigation of the defendant and may not pronounce sentence until it has received a written report of such investi-			୯ନ୍ତ	
	gation.			-	
	further directs that, if a misdemeanor conviction, a pre-sentence			ce	
	report is not required but the court may not pronounce any of the following sentences unless it has ordered a pre-sentence investi-		1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	he	
	gation of the defendant and has received a written report thereof:		an a	ig of:	
			ч.		
	-sentence of probation;				
	-reformatory or alternative local reformatory sentence; -sentence of imprisonment for a term in excess of ninety days;				
	-consecutive sentence of imprisonment for terms aggregating			ays;	
	more than ninety days.			81	
				-	
רסי	200 60			1	and a second second Second second
OF L	390.60	ļ			
. –	directs that, when a person is sentenced to any of the terms of				
	imprisonment mentioned in the misdemeanor section of CPL 390.20				
	or to an indeterminate sentence of imprisonment, a copy of all				
	pre-sentence reports and memoranda be delivered to the person in charge to the correctional facility.				
	TH CHATRE OF OHE COLLECTIONET TECTITORS	1	1	1	

 \odot

CPL

Č

ς,

C

ſ

1

C

C

C

C

C

O

24-

when a defendant is charged with an offense or ess than felony grade, the court must order or bail.

ts that no local criminal court may order recog-il with respect to a defendant charged with a felony til:

trict attorney has been heard or has waived his appear; and

rt has been furnished with a report of NYSIIS ing the defendant's criminal record, if any.

-108-



GLOSSARY

- a report indicating in a concise manner all available information relating to the individual's prior criminal

- a detailed report chronicling all available information relating to the individual's prior criminal justice

- the court or courts which a probation agency serves.

- Criminal Procedure Law -- see Appendix for definition.

- a report transmitted to supervisory agencies (i.e. parole/probation) notifying them that an individual under their supervision has re-entered the justice system via another event.

- a report which indicates to the input agency that the current event represents the individuals initial contact with the system.

	INVESTIGATION	ARRES
OPERATION	In criminal investigation police oftentimes develop a list of suspects who may be responsible for an act. Since the checking out of this list involves valuable time a faster method is desirable.	At this point the a formal charge a their custody. F by law to fingerp suspect and forwa photo to NYSIIS (
SYSTEM ASSISTANCE (PAROLE/PROBATION REPORTING SYSTEM)	The investigating agency will forward its list of suspects to NYSIIS and NYSIIS will search its files and return all available information to the police agency.	NYSIIS will seek individual. If t currently on paro NYSIIS will so in arrest and superv
TYPE RESPONSE	The response will consist of a compre- hensive summary case history on all input suspects.	NYSIIS will respo abbreviated summa to the arresting applicable with a

Ū

()

O

C

D

REST/BOOKING

the police proceed to place se against an individual in Further, they are required serprint and photograph the orward the fingerprints and S (CPL 160.10).

ek to identify the f the suspect is arole or probation, inform both the ervising agencies.

spond with an mmary case history ng agency and where h a notification sory agency.

TABLE 2-2	DISTRICT	ATTORNEY	DECISION	POINTS	AND	REQUIRED	SYSTEMS	ASSISTANCE
-----------	----------	----------	----------	--------	-----	----------	---------	------------

	PRE-ARRAIGNMENT	PRELIMINARY HEARING	GRAND JURY	ARRAIGNMENT
OPERATION	The District Attorney has the right to be heard at this preliminary stage where decisions relating to bail/ROR for felony arrests will be undertaken. (CPL 530.20)	The operations of the district attorney at this stage are similar to those at the arraign- ment stage.	At this point, the district attorney will present the people's case in an attempt to obtain an indictment.	The district attorney upon indictment proceeds in a manner similar to the original arraignment.
SYSTEM ASSISTANCE (PAROLE/PROBATION REPORTING SYSTEM)	In order to efficiently pursue his obligations at this point in the process, the district attorney will require current background information on the individual.	Depending on the status of the accused between arraignment and preliminary hearing (bail/ROR) the prosecutor may required additional information concerning the activities of the defendant subsequent to the pre-arrignment phase.	See Preliminary Hearing.	Depending on the status of the accused (bail/ROR etc.) the prosecutor may require additional updated information to assist his decision-making.
		bre-griftBunkene bugget		
TYPE RESPONSE	NYSIIS will forward an abbreviated copy of the individual's criminal background to the district attorney to assist in the bail/ROR process. This reply will be forwarded to the	If essistance is requested, NYSIIS will reply with either an updated summary case history or a "no further activity" response.	See Preliminary Hearing.	See Preliminary Hearing.
	prosecutor from NYSIIS via the arrest agency as required by law. (CPL 160-40).			

TRIAL

SENTENCE

No Activity.

. .

.

. .

During this stage, the prosecution assumes responsi-bility for presenting the people's case to a jury.

None Required.

If assistance is required at this point, NYSIIS will respond with a full background report to the district attorney.

Although the bulk of systems assistance will have previously been supplied, there exists the possibility that further infor-mation will be required. For example, the prosecutor may need to determine the background of a witness.

None Required.

						• •	
TABLE 2-3 JUDICIAL DE	CISION POINTS AND REQUIRED SYSTEMS ASSISTANCE						
	PRE-ARRAIGNMENT/PRELIMINARY HEARING	ARRAIGMMENT	PRE-TRIAL	TRIAL	PRE-SENTENCE	SENTZNCING	DISCHARGE
OPERATIONS	At this point in the judicial processing, the Court must decide whether or not to grant ball or ROR. The Court will direct that a NYSIIS background history report be forwarded.	At this point (following indictment/information) the court again considers the disposition of the defendant (i.e. ROR, bail, detain, etc.).	In T.O. (Youthful Offender) cases the court requires the probation agoncy to conduct an investigation and document the results in a report. This report is used by the court to determine if the defendant will be given Y.O. treatment during the trial. The court sometimes requires probation to make adult pre-trial investigations in lieu of pre- sentence investigations (see pre-sentence).	No information needed.	Frior to the imposition of sentence, the court requires that the probation agency prepare a pre-sentence report on the convicted individual. (CPL 390.10).	In pronouncing sentence, the court uses the probation report and other reported information concerning the individual. The court could also desire current infor- mation, at this time, as evidence that the defendant has not committed any new offenses while awaiting trial.	The court at this point requires information in order to determine if there exists any cause legal or otherwise for refusing probation release.
SYSTEMS ASSISTANCE (PAROLE/PROBATION REP/JRTING SYSTEM)	NYSIIS assistance at this point will entail forwarding an abbreviated criminal history or no record response to the appropriate court to assist in the preliminary court actions.	Assistance at this point will be available from MISIIS upon receipt of an inquiry from the probation department and will consist of background information.	In most instances, WISIIS assistance at this point would not be necessary due to previous history transmissions. However, if requested, NYSIIS will respond.	None Required.	Upon receipt of the request, MYSIIS will forward a compre- hensive summary case history of the individual to the probation agency preparing the presentence report.	All systems assistance necessary at this stage has been provided in the pre-sentence stage. However, should the court require confirmation that no criminal activity occurred between the pre-sentence and sentence stage,	Upon receipt of a pre-release check inquiry, MYSIIS will sear the individual's files and rep; to the probation agency so that it may prepare its report for the court. However, since NYSI: will routinely be notifying
						NYSIIS will forward either a "no further activity" response or an updated criminal history detailing all events reported to NYSIIS since the last date of transmission.	probation agencies of any new criminal activity on the part of those supervised, assistance here will be minimal.
TYPE RESPONSE	As required by law (CPL 530.20) NYSIIS will respond, upon receipt of a finger- print felony arrest report, with an abbreviated summary case history to the probation office serving the court.	If requested, MISIIS will respond with either a "no further activity" response or an abbreviated summary case history to the probation office for submission to the	The response will be an abbreviated summary case history and will be transmitted to the investigating probation agency.	None required.	See above.	See above.	If required, MTSIIS will respon- with either a "no further activity" response or an abbreviated summary case histor

	ECISION POINTS AND REQUIRED SYSTEMS ASSISTANCE							
TABLE 2-4 PHONATION D			PRE-SENTENCE	TRAMEPER	SUPERVISION	VIOLATION	DISCHARGE	
	PRE-ARRAIGNMENT	PRE-PLEA				• 4 Count 2 CM	PI SCHAMOE	FOLLON-UP DATA
······································		Two types of investigations are made at this point, Y.O. (fowthful	On order from the enset,	Assertance of a transfer case by probation agencies usually requires a background check on the individual.	When a case is received by the probation department, an officer	Probables vielations fall into	Standard procedure at this	Data collected at this point would be
OPERATION	At this point, probation conducts an investigation to determine if a	Offender) eligibility investigations and mouth investigations	probation conducts a comprehensive investigation of the defendant's legal	DECENTIONED CLARK ON THA INGINIUMI.	is assigned and a schedule of association issued to the	three sategories: technical (violating the rules of probation),	point requires a pre-release inquiry on the probationer	used to conduct systems and program effectiveness studies, Probation
	defendant is suitable for ROR (Release on Recognizance) or bail. The investigation consists of an inter-	utilised by the court in determining whether or int the determined	(eriminal) and social history. Results of the investigation		probationer. This period of supervision requires that the	eriminal (committing a new offense,) and absconding. Warrante are usually issued for technical violations and	to assure that there has been no criminal activity on	agencies indicated a need for the acquisition of reliable statistics
	view with the defendant and checks in the community to determine the individual's	is to be afforded T.O. treatmond. Adust interface and the defendant and D.A. on order from the court with the consent of the defendant and D.A. Results of investigations are contained in a report and transmitted	are incorporated in a report and transmitted to the court.	Transfer Out of Bow York State	probation officer have	absconding. Probation requires a	probation.	which would enable them to evaluate the post-probation period of individ-
	residence, employment, and family ties. The datails of this investigation are	to the court.	and transmitted to the court.	The only assistance necessary at this point concerns the need of WIBIIS to be informed of the transfer so it may update its files. Under the	knowledge of all the activities of the probationer.	means of providing the police with adequate information to recognize violators during normal police		usls and groups of individuals. Such information is seen as assisting in
	the primary means by which the arreignment court will determine disposition before			besis system, it was suggested that both the transferring and socoping opensies notify the central reporting system of the transfer. Under		activities.		the evaluation of probation program and supervision effectiveness.
	trial.			the final system, only the accepting agency would notify the central reporting system.	•			
			The sume type of information	Transfer to New York State	Probation moods of this point require a repid report of all	Assistance at this point would estail a procedure which would require the	Probation agencies generally	Decod on the kinds, employeess
STRIM ASSISTANCE	Information mode of this point concern prior grinical history data to both	Detailed information is needed at this desision point to fasilitate probation investigations. The following is the type of data required:	required at pre-plea is required at pre-sentence. In addition,	Prior to accepting an individual for supervision, the New York State	productioners who are arrested. In addition, it was suggested	probation agency responsible for the warrant to notify WYSIIS so that	agreed that to fully implement the proposed system, agencies	and nearser of data reported to
(PAROLE/PROBATION REPORTING SYSTEM)	substantiate information collected in ball/ROR investigations and to	Indexed information indicating if defendant committed a crime	probation would consider it helpful to obtain a listing	agoney would wish to determine whether or not the individual has a eriminal record in How York State.	that the proposed parole/ probation reporting system	WISIIS may post this information to its files. Nowever, due to the	would have to report discharge data to the reporting system.	WISIIS by all agencies concerned with the individual, WISIIS would be able to erecare statistical profiles on
	provide background data to assist the investigator in acquiring additional	while emulting judicial action.	of prior experienced agencies where detailed information is readily		incorporate a central file of all individuals currently on	quantity of warrants which are issued and closed within a short period of	In order to fulfill this probation request, WISIIS	individual protationers so that probation may evaluate its effect-
	information on the individual. System	2. Co-defendant information:	availatle.	Transfer Out of New York State	probation. This could be becamplished via the installation	time, it was deemed advisable that only warrants remaining open for a	proposes to accept inquiries from agencies requesting	Iveness in a particular case. Additionally group tables may be
	seen as significantly aiding the work of the probation officer by providing data	3. Mental Hygiene, Alcoholie and Marcotic Addiction history information.		But to the fast that HTELES will routinely keep supervising agencies inferent of any criminal behavior this information will be available	of a statum file at WYSIIS. This file would contain identification	minimum of two weeks be reported. Based on this dats in its files.	discharge background information. In addition. WYSIIS will	prepared to llustrate the post- probation experience of similarly
	otherwise difficult and time-consuming to acquire.			from the transferring agency. In addition, the transferring agency will possess the pre-sentence response and should make it available	of all individuals currently main- taining an open status in the	WISIIS could recognize a wanted probationer during normal processing.	anticipate receiving pertinent discharge data from each	grouped prior probationers.
				to the receiving agency.	arrestess with outstanding		agency so that it may both update its base files and	
		and the second			dispositions, etc.)		purge the statue files.	
TTPE RESPONS.	WYEIIS assistance at this point will	Depending on the defendant's status between the pre-arraigment and pre-plea phase, the court may sirect that it be formiabed with another	The response at this point will consist of a comprehensive	Trim for to New York State	In the event conditions require a response under the supervision	If a response is distated by a warrant entry on the status file.	NTSIIS will respond with sither a "no further activity"	WTSIIS would, on a pro-arranged basis supply statistical tables to
IIFA MARTINES	entail the forwarding of an abbreviated	cosy of the criminal history; for simple, it the bornhame a new WySIIS	summery case history report.	WTEITS will search its files for any information it has on the	area, MYSIIS would respond with an abbreviated summary case	MTSIIS will respond with a high priority notice to the wanting	response or a comprehensive	either or both individual probation agencies and the State Probation
	probation agency involved in this preliminary investigation.	bail or ROR between the pre-arraignment and pre-plan phases in a star pressure of the plan phase of th	in the second	individual and report either a comprehensive summary case history or a no record response to the prospective New York State agency.	history or a no further activity response.	agency indicating where the individual is being held.		Department. The format of these tables remain to be discussed with
		will respond with either a "no further activity" support aport.		In either case, MYSIIS will be informed of the new supervision area		Additionally, NTSIIS will report this information to the arresting		the new Division of Probation.
		• To be eligible for T.O. Consideration, an offender must fulfill each of		of the probationer so that the files may be updated.		agency so that it is aware of the status of the individual.		and the second
		the following criteria:						and the second second second second
		. be at least 16 and less than 19 years of age.						
	a that is a fair that a start of the start o	, present indictment must be other than a clams "A" felony.						
		, no previous judgment of conviction for a felony.						
	and the second							
				والمحاجب والمناجب والمحاجب والمحاجب والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج	and the second			

and the second second

. Logi en la la la la

TABLE 2-5A LOCAL CORRECTION DECISION POINTS AND REQUIRED SYSTEMS ASSISTANCE

•		DETENTION	POST SENTENCE COMMITMENT	TRANSFER
L O C A L	OPERATIONS	At this point correction authorities are charged with the custody of individuals held prior to arraignment; in lieu of pre-trial bail, awaiting, sentencing, etc.	At this point the facility accepts responsibility for the custody of the convicted individual. It is the responsibility of each facility to notify NYSIIS, via fingerprints, of the receipt of all sentenced individuals.	On occasion, Correction is required to shift inmates between facilities. This shift may be caused by reasons such as over-crowding, br lack of available educational or vocational programs at a
			(Correction Law 618)	particular facility.
C O R E C T I O	SYSTEM ASSISTANCE (PAROLE/PROBATION REPORTING SYSTEM)	NYSIIS will assist these facilities by providing back- ground history, including arrest/ disposition information and when available and officially cert&fied, information relative to mental hygiene admission data; narcotic and/or excessive alcohol Masge, etc. This type of reply is seen as providing the basis for proper super- vision of the inmate.	MTSIIS is required by law (CPL 390.10) to provide information on all persons convicted of a fingerprintable arime. In some cases the replies previously submitted at arraignment or pre-trial will suffice. In either case the report when available must be transmitted to the correction institution (CPL 390.60).	This activity requires no directly system assistance at this time
	TYPE RESPONSE	NYSIIS will respond at this point with an abbreviated summary case history.	When requested by probation agencies preparing pre- sentence reports, NYSIIS will supply comprehensive summary case history. This history will accompany the prisoner to the institution.	None Required.

RELEASE

At this point correction either releases the individual upon reaching maximum expiration of sentence or releases him to parole supervision.

.

Correctional authorities feel there is a need for the complation of statistical studies (both individual and group) which will assist them in the evaluation and planning of institutional programs directed towards the improvement of the inmate.

No system assistance required from NYSIIS at this point. However, NYSIIS will be informed of all releases and type (maximum expiration, parole, etc.) This will be added to the individuals base file and individuals base file and if necessary action will be taken on the status file.

None Required.

System assistance at this point is requested in providing data which would assist the correction adminwould assist the correction admin-istrators in their evaluation process. The depth of NISIIS assistance has not as yet been fully ascertained, however, it is felt that the preliminary steps in this operation will follow one or a combination of the types listed below.

At the present time, NYSIIS envisions reports which may fail into some or all of the following types:

- a. report similar to Uniform Parole Reports;
- b. individual reports at re-arranged intervals to monitor the persons progress following release;
- c. general studies on groups of similar individuals and their progress in the post-release period.

FOLLOW-UP



TABLE 2-5B STATE CORRECTION DECISION POINTS AND REQUIRED SYSTEMS ASSISTANCE

RELEASE

· · · · · · · ·

FOLLOW-UP DATA

See Local Correctional Facilities.

See Local Correctional Facilities

Due to the present involvement of the State Correction Department Research Unit in this type application, no direct system assistance is envisioned. However, NYSIIS will investigate the possibility of providing service in any area as yet uncovered or minimally covered by the department research unit.

At the present time assistance is seen as encompassing those areas mentioned in Section VII of this report.

TABLE 2-5 PAROLE DECI	SION POINTS AND REQUIRED SYSTEMS ASSISTANCE						
	HEARING	RELEASE TO PAROLE	TRANSPER	SUPERVISION	VIOLATION	DISCHARGE	POLLOW-UP DATA
OPERATION	A report is prepared at this point by the Parole Officer showing the social, marital, military and educational background of the potential parolee, This report is one of the factors considered by the Parole Board in granting parole.	Parole is given advance notice when a man will be .eleased and to which area parole office the parolee will be assigned.	Transfer cases are handled by the local transferring and accepting agencies. Transfer reports are forwarded to the Parole Central Office.	At this point, parolees must report to their assigned parole officer at scheduled time periods. Farole officers during this time prepare a report summarizing the parolee's activities during his release.	Parole prepares a report at this time in all instances where a parolee is charged with absconding, a new offense or a technical violation.	Parole will at this point discharge a parolee when his maximum time is up.	Data collected at this point is used to evaluate the effectiveness of parole programs in terms of knowing why some parolees return to the same or different crime patterns that existed when they were institutionalized.
			TRANSPER OUT OP NEW YORK STATE				•
SYSTEM ASSISTANCE (PAROLE/PROBATION REPORTING SYSTEM)	Information required at this point is obtained by the parole officer through inmate interviews, parole area agency investigation and review of the probation report. System assistance at this point would depend on the status of the defendant between conviction and sentence. However, regardless of pre-sentence status, parole will receive a copy of the pre-sentence report from the probation office (CPL 390.60).	Required information at this point is contained in the parole case folder received from Correction. All required information is checked to insure this data is in the folder. If any information is missing a request is made for it through the Department of Correctional Services. When an inmate is released MYSIIS will receive notification of this release and will update its files for the individual.	No assistance is required at this decision point due to the centralised structure of the Division of Parole. All responses on the part of NYSIIS concerning any parole activity are forwarded directly to the Central offices of Parole for action and follow-up. TRANSFER TO NEW YORK STATE New York State upon receiving a request from another state to assume super- vision responsibilities will request background information on the parolee. At present no system assistance is performed, however, NYSIIS does possess the capability of searching its files, based on fingerprints, to ascertain if the individual has a record in New York State. - In either case NYSIIS will be notified of the transfer so that the proper file updating may be performed.	Assistance at this point would be restricted to that outlined in the "violation" decision point.	Assistance may be rendered at this point by providing information on oriminal activities of persons under parole supervision. In processing new arrest information against a status file NYSIIS will be able to recognize whether the individual is a parolee and/or in a wanted status.	No systers assistance is required at this decision point. Parole will notify NYSIIS of all individuals completing parole either through being returned to an institution or satisfactory completion of the supervisory period. NY3IIS will then update its files for the individual.	At present, the Research Unit of the Division of Parole compiles and distributes statistics concerning a wide range of parole operations and parole experience. WYSIIS can assi- this function by providing pertinen background information from its fil In addition, WYSIIS will in the mea future, explore with Parole the possibility of providing active assistance in future statistical programs.
TYPE RESPONSE	If the inmate has remained in custody during the pre-sentence period, there will be no need for NYSIIS to supply any information at this point since the reply given probation for its pre- sentence investigation will suffice. However, if the inmate has been released on bail/ROR during the pre-sentence period, NYSIIS will be prepared to	Noné Required,	TRANSPER OUT OF NEW YORK STATE None Required. TRANSFER TO NEW YORK STATE NYSIIS would reply, with a comprehensive summary case history or no-record response to the Central Offices of New York State Parole.	See Violation	Upon recognizing parole and/or wanted status, NYSIIS will note this condition on the abbreviated criminal history reply to the arrest agency and will immediately notify Parole of the location of the parolec via a high priority notification.	None Required.	Although NYSIIS does not presently respond at this point, information is available in its files, monitori- the current activities of individua released from supervision. Discussions are planned with the Division of Parole which will explo the maximum utilization of this information.
	forward either a "no further activity" response or an updated criminal history report.						

2-7 YOLONE AND MERPORES TIME												· · · · · · · · · · · · · · · · · · ·					
POLICE DECISION POLNTS	V CIL L'HAL/HALE	PORE TIME	DISTRICT ATTORNET DECISION POINTS	TOLO	E/NESPONSE TIME	Phoneficw Bucision Point	TOLINIL/IN			V OLGHE/NES	POINT TINE	CONNECTION ⁴⁵ DECISION = POINT	VOLUNE/10	RIPONSE TIME	PAROLE DECISION POINT	30LINE/101	Pones TDB
	STATE WIDE VOLUME	7114		STATE WIDE	TIME		STATE WIDE TOLUME	708		STATE VIDE VOLUME	TDG	NEW YORK CITY DEPARTMENT	YOLUNK	TIME.		WELDING (10	THE
THA BRITZERIZ TOR		1 to 24 June	PIN-ANIAL CONT	140,000 21	1-2 hours after imput to WTHIN	PIR-ARIALOUMET	Included in pro-	1 to 6 days ()	710-2104240007	140,000 26	1 to 2 hours after impat to WIWIIS	DETENTION SENTENCING {complement	90,000/YR 30,000/YR	2 to 3 heure (8 15 to 30 days (9		18,38:	Bot Applicable
ARMENT/BOOKI IN	140,000 <u>/</u> 1	2 to 4 Jaura		58*000	Not Applicable	PIE-FLEA	Included in pre-	1 to 7 days (5	CREMENT (PINGRABINIANLE	100,000 (Betimate)	Within 3 hrs. after input to WTUIIS	TRANSFER INMATES NOVING IN OR OUT OF AN INSTITUTION	3,500/DAT	No Responde Regulred	NULLAND TO PARCE	3,291	Wet Applicable
			THEMPSELADER	25,000			54,507	1 to b with	PHD-TRIAL MEARING	6,000	1 be 2 days after imput to WYBIIS	PRELEASE	60,090/TR	No Response Registred	TRANSFORM To HTS Deportation	772	Lo Requested
			TRIAL	25,070	Within 1 day	TRANSFER To STD Supervision	2,820	1 to 5 days		40,35 L7		UPON COMPLETION OF SEMTEMORE POLLOW-UP	30,000/7R		To Other State Supervision	679	As Requested
				X/A	Not Applieable	To Other State Supervision	2,808	Calasem	Phil-Ministrat;B	257,520 27	Within 7 days				VIOLATION	19,877 b,274	As Represted
						FOR THE STOR	37,042	As required Desclipto	PELON-OP	257,520 [7	1 to 2 days after input to WINIIS	NEW TORK STATE DEPARTMENT SENTERCING (commitment)	\$,,900	1 month CO	PINCIAN	*,25°	De Response Por
						DIRCHARE		Immediate				TRARSPER NELEASE	7,700	·	FOLLOW-OF	·	
						PULLON-OP				(6 This figure is based on finger- printable arrow for calendar yes 1969. This tota		POLLOW-UP	Lables	Ne strist time requirements		/10 Based on 1969 statistics of	
	ana shina shi Nga sha shi						(7 Based on 1968 statistics /1 No State-wide	A loss to now provi- close requiring ball/NOR investign- tions in felong		will increase with the incep- tion of the new criminal proce-		(8 Due to provisions of new) via response to court and proceedings.	CPL, this information probation agency in	should be available proliminary ball-NCR	•	the Division of Parols,	
	(1 Based on 1969 arrest statistic	ia					stetistics svaliable.	arreignments (CPL 530.20). This time will in many instan- ess be reduced to		dure inv offec- tive 1 Hept 1971 27 This figure in- cludes all court		2 Due to provisions of new eastence and further that inmate to correctional fa	this information be	required prior to transmitted with		an an taon an Taona an taonachta	
								hours, (5 New requirements stipulated in CPL 530.40 effectively		activity finger- printable and no fingerprintable, of a critical							
								530.40 effectively reduce this time estimate.		nature for 1968.			and a final second s				

Agency		Decision Point	Input Type	Response Required	Legal Source	Type Response	How Satisfied
<u></u>	A	Investigation	Inquiry	Yes	None	Comprehensive History or No Record	By Direct NYSIIS Response
Police	В	Arrest/Booking	Update	Yes	CPL 160.10	Abbreviated History or No Record	By Direct NYSIIS Response
	С	Preliminary Arraignment	None	Yes	CPL 160.40 CPL 530.20	Abbreviated History	By NYSIIS Response at B-1
District Attorney	D	Grand Jury	None	No			
	E	Arraignment	None	Yes	CPL 210.15 CPL 530.40	Abbreviated History	By NYSIIS Response at L-1
	F	Trial	None	No			
		4					
	G	Preliminary Arra [†] gnment	Update*	Yes	CPL 530.20	Abbreviated History	By NYSIIS Response at B-1
Court	H	Arraignment	None	Yes	CPL 530.40	Abbreviated History	By NYSIIS Response at L-1
	I	Trial	None	No			
	J	Sentence	Update*	Yes	CPL 390.20	Comprehensive History	By NYSIIS Response at M-1
			2	3	4		

TABLE 4-1 TYPE RESPONSE BY AGENCY DECISION POINT

1 of 2

ADDITIONAL RESPONSES

At the time of response to the Police at the arrest/ booking point (B-1) NYSIIS will also forward responses to the District Attorney, the Court and Probation in order to satisfy their needs at C-1, G-1 and K-1. In addition, should the arrest concern an individual currently under Probation or Parole supervision, NYSIIS will assist Parole and Probation by forwarding a response which will satisfy their needs at N-1 and R-1.

7

None

None

		TABLE 4-1 TYPE RESP(ONBE BY AGENCY	DECISION PO	DINT	and a second	
Agency		Decision Point	Inp ut T y pe	Response Required	Legal Source	Type Response	How Satisfied
	ĸ	Preliminary Arraignment	None	Yes	CPL 530.20	Abbreviated History	By NYSIIS Response at B-1
	L	Arraignment	Inquiry	Yes	CPL 530.40	Abbreviated History	By Direct NYSIIS Response
	M	Pre-Sentence	Inquiry	Yes	CPL 390.10	Comprehensive History	By Direct NYSIIS Response
Probation	N	Supervision	Inquiry/ None* <u>/2</u>	Yes/Yes		Subsequent Report/ High Priority Notice	By Direct NYSIIS Response
	0	Release	Inquiry/ Update*	Yes/No		Subsequent Report/None	By Direct NYSIIS Response
70-90-90 (1999)	2	Classification	Inquiry*	Yes		Comprehensive History	By NYSIIS Response at M-1
	Q	Release to Parole	Update	No			
Maro le	R	Supervision	Inquiry/ None* <u>/2</u>	Yes/Yes	Correction Law 216	Subsequent Report/ High Priority Notice	By Direct NYSIIS Response
	S	Release from Parole	Inquiry/ Update*	Yes/No		Subsequent Report/None	By Direct NYSIIS Response
	Т	Detention	Update	Yes		Abbreviated History	By Direct NYSIIS Response
Correction	U	Admission	Update	No <u>/3</u>	CPL 390.60	Comprehensive History	By NYSIIS Response at M-1
	V	Release	Update	No			
		1	2	3	4	5	6

ADDITIONAL RESPONSES

At the time of response to Probation at the Arraignment point (L-1) NYSIIS will also forward responses to the District Attorney and the Court to satisfy decision points E-1 and H-1.

At the time of response to Probation at the Pre-Sentence decision point (M-1) NYSIIS will also forward responses to the Court, Parole and Correction to satisfy decision points J-1, P-1 and U-1.

If for any reason, Parole does not receive a copy of the NYSIIS report supplied to Probation at the Pre-Sentence decision point (M-1) NYSIIS will, upon request, forward a history report to satisfy the needs of Parole in the classification process (P-1).

If for any reason, Correction does not receive a copy of the NYSIIS report supplied to Probation at the Pre-Sentence decision point (M-1) NYSIIS will, upon request, forward a history report to satisfy the needs of Correction in the admission process (U-1).

7

